



alphanumeric journal

The Journal of Operations Research, Statistics, Econometrics and Management Information Systems



Volume 7, Issue 2, 2019

Received: September 30, 2019	AJ ID: 2018.07.02.ECON.04
Accepted: December 30, 2019	DOI: 10.17093/alphanumeric.640213
Published Online: December 31, 2019	Research Article

# The Nexus of Economic Growth, Trade Openness and Banking Sector Depth In OIC: An Application of Panel Data Analysis

İsmail Durak, Ph.D. \*

Assist. Prof., Department of Quantitative Methods, School of Business, Duzce University, Duzce, Turkey, ismaildurak@duzce.edu.tr

Ergün Eroğlu, Ph.D.

Prof., Department of Quantitative Methods, School of Business, Istanbul University, Istanbul, Turkey, eroglu@istanbul.edu.tr

\* Düzce Üniversitesi İşletme Fakültesi, Konuralp, Beçi Kampüsü 81620 Merkez, Düzce, Türkiye

Ð

Ð

ABSTRACT	This article is investigated the connections between economic growth, trade openness and banking sector depth, using a panel
	data set including seventeen countries in the Islamic Cooperation Organization (OIC), where participation and conventional
	banking co-exist, for the period 1990–2016. Using a multivariate framework, it is primarily found that all the variables are not
	integrated of order one (I). Since the series are not stationary, cross-dependence tests and Westerlund (2007) cointegration
	analysis are performed to the series and it is determined that the series are cross-dependent and cointegrated. Then, the models
	are estimated with three estimators by writing the panel as panel ARDL model to determine the long-term and short-term
	relations. The results of the study indicate a general long-run equilibrium connection between economic growth, trade openness
	and banking sector depth as well as a short-run connection among these variables. Policy suggestions include those that will
	increase greater banking sector depth as well as promoted trade openness.
Keywords:	Economic Growth, Trade Openness, Banking Sector Deoth, Panel Data, Participation Banking



## 1. Introduction

The relationship among economic growth and trade openness is one of the warmest arguments in the literature of growth economics. Mostly many studies on this relationship is based on the thought that trade openness causes economic growth. Empirical studies about relationships among trade openness and economic growth stated that trade openness has an important effect on economic growth (Levine and Renelt, 1992; Greenaway and Sapsford, 1994; Romer, 1998). Broadly, it can be stated that trade openness positively effects the economic growth in four ways: efficiency of comparative advantage effects, the multiplier of foreign trade effect on real production, improvement in foreign exchange markets, and accelerated capital formation and technical change (Kugler, 1991; Reppas and Christopoulos, 2005). On the other hand, banking sector depth itself may be connected to trade openness and economic growth. It could affect economic growth directly through the usual expenditure channels while indirectly through its impact on trade openness. There are some in the literature examining the relationship between economic growth and banking sector depth (For Example, Kar et al., 2011; Gries et al., 2009; Ang and McKibbin, 2007; Dritsakis and Adamopoulos, 2004; Craigwell et al., 2001; Ahmed and Ansari, 1998; Greenwood and Smith, 1997). However, in these studies, a common conclusion could not be reached about the relationship between the related concepts. Some of the countries in the sample discussed in the study are oil exporting countries and their economic growth is mostly based on oil exports. These countries aim to diversify their economic growth by encouraging the development of other economic sectors since the 1980s (Al-Moulani, 2016). In this respect, participation banking can be one of the important alternative channels to achieve this. Furthermore, it is important to examine the long- and short-term relationship between trade openness and economic growth within the framework of the determined models.

In this study, rather than working with just time series data or taking only crosssectional units, panel data is used which give more and healthy information than other methods and explain the relationship between the variables. Also, this study is one of the rare studies examining the relationship between related variables, especially in terms of the sample discussed. Moreover, the limited number of studies examining these three concepts together is another original aspect of the study. In addition, the use of the banking sector depth composite index variable to represent financial depth in a broad sense is another point that makes the study unique. On the other hand, examining the relationship between the related variables by using panel data analysis, which is one of the strong econometric methods, and looking at the short and long term relationship between the variables by using panel ARDL method is one of the features that make the study different from the similar ones. The representation of the relationship between economic growth, banking sector depth and trade openness is presented in Figure 1.





**Note:** BSD is the banking sector depth index constructed from DCB, DCP, BRM, CLP BSI; GDP is per capita economic growth; TO (trade openness). All the variables are defined in Table 3.

**Figure 1.** The structural framework on the possible linkages between banking sector depth, economic growth, and trade openness. Sources: Pradhan et al. (2017a), Puryan (2017).

In this study, we investigate to answer questions related the nature of the relationship between economic growth, trade openness and. banking sector depth. The novel features of this paper are that: (1) we use a group of 17 OIC (Organization of Islamic Countries) countries over a long time, from 1990 to 2016; (2) we combine a broad scope of the literature; and (3) we apply principal component analysis, first and second generations of unit root tests, cross-dependence tests, Westerlund (2007) cointegration test, PMG, MG and DFE estimators to test panel ARDL models. These formulations are rarely employed in the finance-growth literature. The remainder of the paper is structured as follows: Section 2 provides an overview on two sides of the economic growth literature: one examining the relationship between banking sector depth and economic growth, and the other seeking trade openness and economic growth. This part also motivates the study by indicating the unique contributions of the current research. Section 3 gives a brief summary on conventional and participation banks. Section 4 describes the variables in more detail and presents the data source used in the analysis. Section 5 outlines the empirical econometric model and estimation strategy, and this is continued by section 6 which is showing the results. The final section contains a summary and the policy implications of our results.

## 2. Literature Review

## 2.1. Relationship Between Economic Growth and Banking Sector Depth

The relationship between finance and growth begins with Bagehot's (1873) articles on classical thought and later with the work of Schumpeter (1912). On the other hand, modern literature on economic growth often begins with research that led Robert Solow to receive a Nobel Prize in the mid-1950s. Nevertheless, the theoretical and empirical literature of this period has focused mostly on the role of capital and labor resources and the use of technology as growth resources to ensure economic growth. Therefore, the role of the financial sector in the growth process has been ignored until



the 1970s (Wachtel, 2001). Conceptual and empirical studies on the relationship between finance and economic growth have increased, especially as some important economists such as Goldsmith (1969) and McKinnon (1973) draw attention to the relationship between financial structure and banking to economic growth. Thus, in the last quarter of the century, many studies have been conducted both theoretically and empirically using various data sets to investigate the relationship between financial development and economic growth, The findings generally provide evidence to the view that financial development, and particularly the development of the banking sector, supports economic growth (e.g., Beck et al., 2000; Levine et al., 2000). Although the relationship between financial development and economic growth has been discussed for many years, there is still no common judgment There are basically four opinions on the subject. The first view is the supply-leading view that means "financial development promotes economic growth and ensures economic growth". The second view is that financial development only follows economic growth and its role in achieving economic growth is exaggerated. In short, the second view is demand following hypothesis that the concept of "economic growth leads to financial development". In addition to the above two hypotheses, a third view is that of those who argue that economic growth and financial development can complement each other. According to these, there is a bilateral causality between economic growth and financial development (Greenwood and Smith, 1997). According to proponents of this hypothesis, financial development is indispensable to economic growth, and good economic growth inevitably requires a well-functioning and efficient financial system. The fourth view is that of those who argue that financial development and economic growth can develop independently from each other and therefore there is no causality between them (Chandavarkar, 1992). As the Finance-Growth literature both expanded and developed, complex models emerged from the early 1990s on the relationship between financial development and economic growth. As the studies of Greenwood and Jovanovic (1990), King and Levine (1993b), Pagano (1993), Bencivenga et al. (1995), Greenwood and Smith (1997), Blackburn and Hung (1998), various techniques have been used to model the connection between financial development and economic growth. Some of the new findings point out that the link between finance and growth is not linear, so it is suggested that the relationship between banking sector depth and economic growth has become negative after a certain level (Huang and Lin, 2009; Arcand et al., 2012; Barajas et al., 2013a). Judging from the empirical studies, empirical studies on the relationship between financial sector and economic growth have been shaped from King and Levine's (1993a) research on post-war countries in the 1990s, and from Wachtel and Rousseau's (1995) long time series for several countries. Later, there has been a huge increase in studies on financial depth and economic growth. In particular, the 2008-2009 Global Financial Crisis has enabled us to closely examine not only the global financial system and economy, but also the studies in the fields of finance and economy. One of the first studies to investigate the relationship between financial development and economic growth is Goldsmith's (1969) study. In this study, data from 35 countries between 1860 and 1963 were analyzed in an empirical model. As a result of this study based on the OLS (Ordinary Least Square) model, it was stated that an above-average financial development (represented by the ratio of financial intermediation assets to gross national product) was accompanied by high economic growth periods. Along with the financial development representing banking variables, new studies have



been conducted examining the contribution of stock markets to economic growth with the development of stock markets. For example, Atje and Jovanovic (1969) applied the OLS technique using the annual observations of 94 countries between 1960-1985. As a result, there have been some conclusions that stocks markets had positive effects to the economic growth. On the other hand, Barro (1991) and King and Levine (1993a, b) 's work on the relationship between finance and growth has been the trigger for studies with cross-country data sets. Barro (1991), in the study of 98 developed and developing economies in the 1960-1985 period, used GDP per capita and some human capital variables. The study was concluded that the growth rate of GDP per capita is positively related to human capital and negatively correlated with the initial level of GDP per capita. One of the recent cross-sectional regression studies is Beck (2011). The author examines the finance-growth relationship in resource-based economies to determine whether there is an abundance dilemma in financial development. In this study, the ratio of private sector loans to GDP, the ratio of liquidity debts to GDP and some natural resources were selected as financial development proxies. As a result of the study, it was concluded that there is no significant difference between natural resources-based economies and financial development compared to other countries. Secondly, empirical studies based on time series analysis examining the relationship between finance and growth were examined. In these studies, mostly vector autoregressive (VAR) technique, Granger causality tests, multivariate cointegration tests techniques were used. Jung (1986) applied the Granger causality tests to the data of the period of 1950-1981 belonging to 56 countries. The narrow money (M1) and the broad money (M2) variable, were used as two alternative financial development variables. The results supported the "finance supports growth" approach, which is the supply-leading view. In addition, in the studies of Rousseau and Sylla (2003), the researchers re-confirmed the approach that financial development supports economic growth by using the data of 17 countries between 1850 and 1997. Moreover, Rousseau (1999) applied a Meiji period in Japan in 1868-1884 using VAR procedures in a time series study on a single country. As a result, it is concluded that the financial sector serves Japan's explosive growth. Mohamed (2008) examined the impact of financial development on economic growth in Sudan between 1970 and 2004. The short-term and long-term relationship between financial development and economic growth was estimated using the autoregressive distributed lag (ARDL) cointegration approach developed by Pesaran and Shin (1999). ARDL results indicate cointegration between variables. Accordingly, it was found that there was a positive but statistically insignificant relationship between the ratio of broad money supply to GDP and economic growth. In addition, a negative and statistically insignificant relationship is found between the ratio of private sector loans to GDP and economic growth. In summary, the author concludes that financial development indicators do not have a direct impact on real economic growth. This is due to the inefficient allocation of resources by banks, the lack of a suitable investment environment necessary to promote significant private investment in the long run, and the poor credit quality of the banking sector. Although the time series studies have increased and enriched the financial-growth literature, they have serious problems arising from short estimation periods, especially due to limited data. In other words, the use of short time series prevents reliable time series analysis because it requires long time series to appropriately calculate the link between variables and effective dynamics. To cope with the degree of freedom, many



studies describe only a lag in the empirical model specifications. This gives serial correlation problems and / or poorly defined models. Another commonly known problem with time series studies is the misinterpretation of Granger causalities. In Granger causality tests, if the lagged values of one variable help to predict the present value of another variable, it is therefore not correct to say that there is definitive evidence of the cause-effect relationship. Among some panel data studies, Benhabib and Spiegel (2000) examined whether financial intermediation development affects economic growth, investment and total factor productivity increase by using panel data covering 1965-1985. In the study conducted using the GMM panel estimator, financial development indicators are found to be associated with both total factor productivity increase and accumulation of both physical and human capital. What differentiates the study from its peers is the various variables they use for financial and economic growth. Loayza and Ranciere (2004) examined the finance-growth relationship through a panel error correction model derived from the panel ARDL (autoregressive distributed lag model). As an alternative to traditional time series methods, pooled mean group (PMG) estimator of Pesaran, Shin and Smith (1999) is used to find long- and short-term effects between variables. As a result, it was found that there is a long-term positive relationship between financial intermediation and growth, but the study also concludes that there is a short-term but negative relationship between these variables. Furthermore, the study concluded that the positive relationship between long-term economic growth and financial development is less in countries affected by the banking crisis than in countries not affected by the crisis. Law and Singh (2014) was examined the relationship between the development of the financial system and economic growth by using dynamic panel data analysis using 1980-2010 period of 87 developed and developing countries data. In the study, the three main variables used for financial depth (ratio of private sector loans to GDP, ratio of liquidity liabilities to GDP, ratio of domestic loans to GDP) and various control variables are used to represent economic growth in the literature representing the development of financial system. As a result of the study, it was concluded that there was a threshold point in the relationship between economic growth and finance, and financial depth in economies below the threshold point will positively affect economic growth. In addition, there is some evidence on that the economies above the threshold, financial depth will adversely affect economic growth and that financial depth is not always good for economic growth, and even after a certain threshold, it is detrimental to economic growth. Aliu and Abazi (2015) were investigated whether financial depth had a significant effect on economic growth by using the annual data of 7 Western Balkan countries in the period 1980-2014. In the study carried out by taking various variables related to depth of financial sector, broad of financial sector and quality of financial sector as a criterion of financial deepening, the effects of these variables on economic growth are estimated by using panel data analysis. The findings are different from the expectation that financial deepening accelerated economic growth. In fact, conclusions have been reached in line with the findings of recent studies emphasizing that more than a certain level of financial deepening may turn into a disadvantage for economic growth. Apart from these studies, some of the other cross-sectional, time series and panel data studies related to the subject are given in Table 1.



Studies	Sample	Data	Method	Result
Goldsmith (1969)	35 Country	1860-1963	Cross- Section	There is a positive and significant relationship between financial development and growth.
Levine (1991)	49 Country	1960-1990	Cross- Section	Liquidity in financial markets facilitates long-term investments and increases productivity by increasing productivity.
Hermes &Lensink (2003)	67 developing country	1970–1995	Cross- Section	It is stated that a certain level of financial development is a prerequisite for obtaining growth benefits from foreign direct investment.
Ghali (1999)	Tunisia	1963-1993	Time series	It is confirmed that financial development is the cause of economic growth.
Neusser &Kugler (1998)	14 OECD country	1970-1991	Time series	For the size of the financial system, if the value-added measures provided by the financial system are used instead of simple criteria, the effect on economic growth will be positive and strong.
Arestis et al. (2001)	France, Germany, Japan, United Kingdom, United States	1973-1997 1972-1998 1974-1998 1968-1997 1974-1998	Time series	The study is concluded that banks are more powerful in supporting economic growth than stock markets.
Boulila&Trabelsi (2004)	16 Country	1960-2002	Time series	In nine of the fifteen countries included in the study, there is a long-term relationship between financial development and economic growth.
Nili&Rastad (2007)	12 Petroleum Exporting Country	1975-2000	Panel data	The interaction between the development of financial intermediation and investments is negatively associated with economic growth in highly oil- dependent countries.
Kar et al. (2011)	15 MENA Country	1980-2007	Panel data	It is stated that financial sector development does not support economic growth in MENA region. The findings confirm the source-based demand following approach.
Abu-Bader&Abu- Qarn (2008)	Algeria, Egypt, Israel, Morocco, Syria, Tunisia	1960-2004	Time series	The results confirm the approach that financial development supports long-term economic growth.
Bhattacharyya&Hodler (2014)	133 Country	1970- 2005	Panel data	It is stated that strong and democratic political institutions support financial development in a resource rich economy.

**Table 1.** Some studies on Economic Growth and Banking Sector Depth

## 2.2. The Relationship of Economic Growth and Trade Openness

The relationship between economic growth and trade openness is one of the most current debates in the field of development economics. Many of the study results on this relationship are based on the approach that trade openness provides economic growth. In the literature, this approach is called the "trade-oriented growth" hypothesis (Giles and Williams, 2000; Reppas and Christopoulos, 2005). It is stated that trade openness has the role of a locomotive for the growth of the real economy along with many benefits other than the productivity it provides (Manteli, 2015). On the other hand, it could be better if less developed countries orient their development towards an output expansion for their domestic market. In the theoretical context, Adam Smith (1937) and David Ricardo (1973) first confirmed the positive relationship between trade openness and growth. According to the Smith and Ricardian model, countries increase their per capita income by specializing in the field in which they have comparative labor-productivity advantages (Pigka-Balanika, 2013). This approach is called "comparative advantage theory". On the other hand, Walter et al. (2012) stated that theories examining the relationship between trade openness and macroeconomic variables can be classified into four approaches: Keynesian income approach, flexibility approach, absorption approach and monetary approach



(Dornbusch, 1975; Johnson, 1977). The Keynesian approach suggests that growth with domestic capital plays a key role in trade openness relative to foreign capital, while the flexibility approach emphasizes the importance of exchange rate in determining trade openness. Moreover, the absorption approach argues that the increase in economic growth increases the trade openness, while the monetary approach plays an important role in the rapid increase in the money supply. Based on these four approaches, it can be said that trade openness is mostly related to economic growth and other macroeconomic common variables. Although there are many studies suggesting that trade openness will positively affect economic growth based on endogenous growth theory, on the other hand, based on the Romer (1990), Grossman and Helpman (1990), Rivera-Batiz and Romer (1991), Matsuyama (1992), Yanikkaya (2003) stated that trade constraints can reduce the growth rate around the world. Although many different models and theories have been proposed for the link between trade openness and economic growth, as seen in conceptual literature studies, this relationship is still not fully elucidated. On the other hand, the view that trade openness supports economic growth is supported by many empirical studies. According to some empirical studies; for example, Dollar (1992), Edwards (1998), Frankel and Romer (1999), there is a positive relationship between trade openness and economic growth. On the other hand, Rodriguez and Rodrik (2000) stated that this effect will vary according to the methodology and preferred proxy for trade openness. Abbas (2014), also, argued that trade openness have a negative impact on economic growth. On the other hand, Srinivasan and Bhagwati (2001) claim that Rodriquez and Rodrik (2000)'s criticism of the positive relationship between economic growth and trade openness is not sufficiently convincing and evidencebased. In this context, the literature on the relationship between economic growth and trade openness can be divided into three parts in a broad sense. These are crosssectional analysis studies, time series studies and panel data studies. Edwards (1998) examined the relationship between trade openness and total factor productivity with the method of cross-sectional analysis by taking 98 country data into account. In this study, nine different indices are used to investigate whether there is a generally accepted positive relationship between these indices and economic growth. According to the results of the study, more open economies have grown faster than other economies. Sachs and Warner (1995) investigated the relationship between trade openness and economic growth with the cross-sectional data of 122 countries based on production function. Although researchers found some evidence that economic growth will be positively affected by trade openness, they stated that it is not enough to produce growth. Along with trade openness, they were concluded that there is a need for stable macroeconomic policies, structural policies and institutions in order to achieve economic growth. Studies using cross-sectional data have been criticized for some problems. One of them is the pseudo-correlation problem that arises from the fact that it is not stationary in the cross-sectional data. Another problem arising from the use of cross-sectional data is that it is not allowed to examine the causality aspect of variables conducted with such kind of data (Christopoulos and Tsionas, 2004). In their study, Hansson and Jonung (1997) examined the long-term relationship between financial development and economic growth with cointegration analysis, one of the time series analysis methods, using Sweden's data from 1830 to 1990. The empirical study shows that there is an interaction between the studied variables and that the estimated contribution of the



financial system to economic growth is highly dependent on the time period of study and the variables used. Yapraklı (2007) examined the interaction between the economic, financial and trade openness on economic growth in Turkey. In the study conducted using data from 1990-2006 period, cointegration, causality and vector error correction tests, one of the time series methods, were tested. According to the empirical results, economic growth was positively affected by trade openness and negatively affected by financial openness in the long term. In addition, the causality tests created by vector error corrections were found to be a two-way causality between economic growth and financial and trade openness. However, it was concluded that there is another unilateral causality from trade openness variable to the financial openness variable. Solarin and Shahbaz (2015) investigated the relationship between economic growth and trade openness through the ARDL boundary approach, cointegration analysis, and granger causality tests, using the 1971-2002 annual data of the Malaysian economy. The results of the causality analysis were feedback between the series, so there was a variety of evidence that economic growth promotes economic growth as well as trade openness. Edwards (2004) examined the impact of trade openness and financial openness on economic growth performance using data from 157 countries for the period 1970-2001. In this study using panel data, He found that the countries that are more commercially open have a lower growth tendency than the countries with a lower degree of trade openness. On the other hand, there were some conclusions that the negative impact on the economic growth resulting from financial openness is reduced through trade openness. Kök et al. (2010) investigated the relationship between trade openness and economic growth with a production function established using data from 1971-2002 period of 51 developed and developing countries. In the study using panel cointegration estimation methods, [trade volume to GDP ratio \* (country population / world population)] was taken as a measure of trade openness. According to the results, the finding that trade openness was an important factor hindering economic growth in less developed countries was found to be significant. This was in parallel with the "impoverishing growth" approach expressed in the literature. On the other hand, one of the other results was that trade openness is a factor that increases economic growth in developed countries. Apart from these studies, some other studies are given in Table 2.



Studies	Sample	Data	Method	Result
Levine& Renelt (1992)	119 Countries	1960-1989 1974-1989	Cross- Section	It was stated that there is a relationship between trade openness and economic growth and its existence may be due to the increase in resource accumulation, not the better allocation of resources.
Arteta et al. (2001)	61 Countries	1973-1981 1982-1987 1988-1992	Cross- Section	There was some evidence to show that liberalization in capital movements encourages growth.
Bahmani& Niroomand (1999)	59 Countries	1960-1992	Time Series	It was concluded that there is a long-term positive relationship between trade openness and economic growth.
Ghatak et al. (1995)	Turkey	1955-1990	Time Series	In accordance with the internal growth theory, it was found that there is a stable, common long-term relationship between real GDP per capita and trade liberalization, human and physical capital.
Chang et al. (2013)	9 Provinces of South Africa	2013	Panel data	There might be evidence that the relationship between economic growth and exports was varying from province to province, and that expansion of exports may not be an effective strategy for achieving economic growth in South Africa.
Yanikkaya (2003)	100 developed &developing countries	1970- 1997	Panel data	Several findings have been showed that trade opening, along with trade barriers, has a positive effect on growth.
Sarkar (2008)	51 Countries	1981- 2002	Panel data	Various evidence has been found to suggest a positive relationship between trade openness and economic growth in rich countries.

Table 2. Some studies on economic growth and trade openness

#### 2.3. Conventional and Participation Banking

There are various types of banks in the world and among these banks, the two most striking types are conventional and participation banking. Conventional banking can be defined as all banks operating on the principle of interest. Participation banking, on the contrary, refers to the types of banking that are established based on Islamic principles and principles that carry out their financial activities without interest. As a field of activity, conventional banks collect funds, act as intermediary between customers, provide loans, support the implementation of monetary and credit policies, provide financial support to investors and businesses, become partner in projects, buy and sell various securities and safequard customers' securities in safe deposit boxes. Conventional banks also have services as actively participating in stock and stock activities, supporting the development of the country through various investments and projects. Participation banks are one of the financial institutions that perform banking activities according to Islamic procedures. As it is understood from the definition, these banks should carry out all kinds of activities within the framework of Islamic principles. Participation banking has succeeded in attracting customers who do not wish to use conventional banking, which is based on interest, especially in their financial business, thus making a significant contribution to the economy by activating non-circulating resources. Interest-free banking has become increasingly widespread not only in the Middle East but also in other countries such as the United States and the United Kingdom. With the introduction of the interest rate prohibition in Islam, credit societies and cooperatives, which have been working without interest in many Muslim countries since then, have gained a different dimension with the emergence of the first example of interest-free banking



institutions in the early 1960s. Between 1963 and 1967, an attempt was made to establish Islamic principles for the realization of financial relations in the Egyptian city of Mit-Ghamr. The Mit-Ghamr initiative, modeled with German savings banks, is circulated small savings in rural areas largely through savings accounts. In this attempt, no interest is paid to account holders, but as an incentive, they are entitled to receive small short-term interest-free loans for investment purposes, and account holders can withdraw their deposits on demand. In addition, investment accounts on Mudaraba basis are introduced. The funds mobilized in this way are based on profit sharing and loss sharing with entrepreneurs (Zaher and Hassan, 2001).

## 3. Variables and Data Structure

Variables that would represent the banking sector depth, economic growth and trade openness determined by the literature and Pradhan et al., (2017a) study. In particular, it is important to consider the variables of banking sector depth in a broad context in order to reflect the depth of the sector sufficiently. As many previous researchers have pointed out (eq Gries et al., 2009; Pradhan et al., 2017a; Karabıyık and Taşkın, 2016; Naceur and Ghazouani, 2007; Rousseau and Wachtel, 1998; Levine and Zervos, 1998; Abu-Bader and Abu -Qarn, 2008; Beck and Levine, 2004) depth of banking sector variable may not be indicated by a single criterion (proxy) or variable. In this study, five variables were used to represent the depth of the banking sector. These variables are domestic credit provided by the banking sector (DCB), domestic credit to the private sector (DCP), broad money supply (BRM), claims on the private sector (CLP) Composite index of banking sector depth (BSI). The last variable (BSI) is a composite index and is derived from four other variables using principal component analysis. The definition of the first four variables taken from the World Bank is detailed in Table 3. This table also describes the other variables used in the study. Other variables used in the study are the annual growth rate of per capita income representing economic growth (GDP) and the ratio of trade openness to gross domestic product representing trade openness (TO). As Harrison (1996) states, the ratio of exports plus imports to gross domestic product is a fundamental and widely used indicator of trade openness. The conceptual framework of the study is shown in Figure 1 for clarity.

Variables	Definitions
DCB	Domestic credit provided by the banking sector: This includes all credit to various sectors on a gross basis, except for credit to the central government, which is net. The banking sector includes monetary authorities and deposit money banks, as well as other banking institutions such as building loan associations and mortgage. This variable is expressed as a percentage of gross domestic product.
DCP	Domestic credit to the private sector: This refers to financial resources provided to the private sector by financial corporations, such as through loans, purchases of nonequity securities, and trade credits and other accounts receivable, that establish a claim for repayment. This variable is expressed as a percentage of gross domestic product.
BRM	Broad money supply: This is the sum of currency outside banks; demand and term deposits, including time, savings, and foreign currency deposits of resident sectors (other than the central bank); traveler and bank's checks; and other securities such as trade paper and certificates of deposit. This variable is expressed as a percentage of gross domestic product
CLP	Claims on the private sector: This includes claims on central government that cover loans to central government institutions (net of deposits) -stated as a percentage of gross domestic product.
BSI	Composite index of banking sector depth: This is obtained using four indicators: domestic credit provided by the banking sector, domestic credit to the private sector broad money supply, claims on the private sector. This index is obtained using principal component analysis. The four indicators are defined in this table.
ТО	Trade openness: The sum of exports and imports of goods and services (total volume of trade) measured as a percentage of gross domestic product.
GDP	Growth rate of per capita income (in percentage): Income is called as gross domestic product (GDP). It is the measure of economic growth.
	Course: https://data.worldbapk.org/products/wdi

#### Table 3. Definitions of Variables

Source: https://data.worldbank.org/products/wdi



The data of the study is obtained from World Development Indicators (WDI) website and the UN statistical database (https://unstats.un.org/home/). In this study, which is based on data from the period 1990-2016, the countries included in the analysis consist of 17 members of the Organization of Islamic Cooperation, which has a dual banking (conventional and participation banking) system. In the study, these seventeen countries are abbreviated as All Countries Group (ACG). These countries are Algeria, Bahrain, Bangladesh, Egypt, Indonesia, Jordan, Kuwait, Lebanon, Malaysia, Pakistan, Qatar, Saudi Arabia, Senegal, Gambia, Tunisia, Turkey, United Arab Emirates, over the period 1990–2016. Also, it is aimed to carry out the research in a much wider scope by considering the banking sector depth and stock sector depth which are two dimensions of financial depth. However, due to the fact that the stock markets of some countries were established very recently (for example the United Arab Emirates) and some countries do not have a sufficiently long time series (Naceur et al., 2014), the study was carried out only by taking into account the variables of banking sector depth. On the other hand, due to the fact that a significant part of the financial depth or development and the basis of the financial market is based on the banking sector more than the other countries (Aliyu et al. 2017; Hussain et al., 2015; Kammer et al., 2015) in this sample, the findings are of great importance in terms of reflecting the overall financial system in the sample.

## 4. Econometric Modelling and Estimation Strategy

The following model specification outlines the link between the GDP growth rate, trade openness and banking sector depth in this study;

$$GDP_{it} = f(TO, BSD)$$
(1)

GDP=GDP growth per capita, TO = trade openness, BSD = variables of banking sector depth. The term BSD have five different banking sector depths variables representing (DCB, DCP, BRM, CLP and BSI) into the model. Each of these variables included the model (while other variables stay constant) respectively. If the variables of the banking sector depth are expressed more clearly, the five basic model specifications used in the study can be shown as follows;

 $GDP_{it} = f(TO, DCB)$ <sup>(2)</sup>

$$GDP_{it} = f(TO, DCP)$$
(3)

$$GDP_{it} = f(TO, BRM)$$
(4)

$$GDP_{it} = f(TO, CLP)$$
(5)

$$GDP_{it} = f(TO, BSI)$$
(6)

TO= trade openness, DCB= domestic credit provided by the banking sector, DCP= domestic credit to the private sector, BRM= broad money supply, CLP= claims on the private sector, BSI= Composite index of banking sector depth.

One of the methods used to analyze the relationships between panel data variables is cointegration analysis. Cointegration analysis determine whether the variables in the series move independently or dependently in the long run. If there is a cointegration relationship between the series, this means that the deviation from the



existing long-term relationship between the variables is not permanent but temporary and that the error correction function corrects these deviations and converges to the long-term relationship (Uslu, 2012). The two traditional techniques used to test for cointegration between variables are the Engle and Granger method and Johansen technique. The Engle and Granger method is a one-equation technique and can therefore lead to conflicting results, especially if there are two integrated variables. Also, in Johansen technique, if there are more than one cointegration vectors, it is often difficult to interpret each economic relationship and find the most suitable vector (Ang, 2010). In addition, since the validity of both Engle-Granger and Johansen techniques are dependent to first-order stationary of variables, these techniques are criticized (Samargandi, 2015). Therefore, if the variables have a mixed stasis at the level of I (0) and I (1), the two cointegration techniques cannot be used. In this study, Westerlund (2007) error-based cointegration and Pesaran, Shin and Smith (1999) panel cointegration (ARDL) technique called autoregressive distributed lag model approach was used. The error correction panel cointegration test proposed by Westerlund (2007) is one of the important tests used to test for cointegration. This method developed by Westerlund (2007) uses four tests to determine whether there is cointegration or not. Two of these tests (Gt and Ga) show group statistics and two of them (Pt and Pa) show panel statistics. While group statistics provide inference for the units in the panel, panel statistics make inferences for the whole panel. The basic logic in this method is to test for cointegration by determining whether there are error corrections for individual panel members or for the entire panel. This method is one of the most suitable cointegration test methods that can be used in case of unit root and cross-sectional dependence (Gautam and Paudel, 2018). Error correction based cointegration tests are very flexible and allow heterogeneous determination of both long- and short-term specifications of the error correction model (Westerlund, 2007). The panel ARDL model approach is testing the existence of a cointegration relationship between the series without requiring equal integration of series. In general, the autoregressively distributed lag ARDL (p, q) (prepresents the lag of the dependent variable and g represents the lag of the independent variable) can be expressed as follows;

$$Y_{it} = \sum_{j=1}^{p} \lambda_{ij} Y_{i,t-j} + \sum_{j=0}^{q} \delta_{ij} X_{i,t-j} + \mu_{i} + \varepsilon_{it}$$
(7)

Here, i = 1,2, ..., N is the total number of countries, t = 1,2,... T time series in the series,  $\mu_i$  constant effects, j = the number of lags, X<sub>it</sub> independent variables vector (kx1), Y<sub>i,t</sub>j dependent the lagged value of the variable,  $\delta_{ij}$  (kx1) coefficients vector and  $\lambda_{ij}$  the coefficient of lags of the dependent variable.

In the above equation (7), the variables at the level can be re-arranged after grouping and expressed as error correction equation as follows (Mamun et al., 2013: 570);

$$\Delta Y_{it} = \mu_i + \phi_i (Y_{i,t-1} + \theta'_i X_{i,t}) + \sum_{j=1}^{p-1} \lambda_{ij}^* \Delta Y_{i,t-j} + \sum_{j=0}^{q-1} \delta_{ij}^{**} \Delta X_{i,t-j} + \varepsilon_{it}$$
(8)

Here,  $\phi_i = -(\beta_i / \phi_i)$  refers to the long-term or equilibrium relationship between Yit and Xit.  $\lambda_{ij}^*$  shows the previous-term coefficients of the dependent values in the model and  $\delta_{ij}^*$  is the short-term coefficients of the lagged independent variables. However, the error correction coefficient  $\phi_i$  is the measurement value of the convergence rate of Yit to the long-term equilibrium value following the change in the independent variable (Xit). It can be said that there is a long-term relationship if the negative value of  $\phi_i$  is met ( $\phi_i < 0$ ). Therefore, if the coefficient  $\phi_i$  is negative and



statistically significant, it proves that there is a cointegration relationship between  $Y_{it}$  and  $X_{it}$ . Since the main purpose of this study is to examine the depth-growth-trade-openness relationship of the banking sector, ARDL and error correction modeling is fully consistent to determine the long-term relationship and short-term dynamics between banking sector depth, economic growth and trade openness.

In a panel data specification, the basic model specification in this study shown in Equation 1 can be shown as the panel ARDL model as follows;

$$\ln GDP_{t} = \beta_{0} + \sum_{j=1}^{p} \lambda_{i} \ln GDP_{t-i} + \sum_{j=0}^{q_{1}} \delta_{j} \ln TO_{1t-i} + \sum_{j=0}^{q_{2}} \varphi_{l} \ln BSD_{2t-i} + \varepsilon_{t}$$
(9)

InDP is the logarithm of GDP growth per capita; InTO is the logarithm of trade openness and InBSD is a set of depth determinants that includes banking sector variables (DCB, DCP, BRM, CLP, BSI).

The five basic models (by writing 5 different variables of banking sector depth individually in this study), which are formed as an error correction model (ECM) of panel ARDL model given by equation (9) can be expressed as follows;

Model 1.

$$\Delta \ln GDP_{it} = \beta_0 + \beta_1 \ln GDP_{t-1} + \beta_2 \ln TO_{t-1} + \beta_3 \ln DCB_{t-1}$$
$$+ \sum_{i=1}^{p} \lambda_i \Delta \ln GDP_{t-1} + \sum_{j=0}^{q_1} \delta_i \Delta \ln TO_{t-j} + \sum_{l=0}^{q_2} \varphi_l \Delta \ln DCB_{t-l} + \varepsilon_{it}$$

Model 2.

$$\Delta \ln GDP_{it} = \beta_0 + \beta_1 \ln GDP_{t-1} + \beta_2 \ln TO_{t-1} + \beta_3 \ln DCP_{t-1}$$
$$+ \sum_{i=1}^{p} \lambda_i \Delta \ln GDP_{t-1} + \sum_{j=0}^{q_1} \delta_i \Delta \ln TO_{t-j} + \sum_{l=0}^{q_2} \varphi_l \Delta \ln DCP_{t-l} + \varepsilon_{it}$$

Model 3.

$$\Delta \ln GDP_{it} = \beta_0 + \beta_1 \ln GDP_{t-1} + \beta_2 \ln TO_{t-1} + \beta_3 \ln BRM_{t-1}$$
$$+ \sum_{i=1}^{p} \lambda_i \Delta \ln GDP_{t-1} + \sum_{j=0}^{q_1} \delta_j \Delta \ln TO_{t-j} + \sum_{l=0}^{q_2} \varphi_l \Delta \ln BRM_{t-l} + \varepsilon_{it}$$

Model 4.

$$\begin{split} \Delta \ln GDP_{it} &= \beta_0 + \beta_1 \ln GDP_{t-1} + \beta_2 \ln TO_{t-1} + \beta_3 \ln CLP_{t-1} \\ &+ \sum_{i=1}^p \lambda_i \Delta \ln GDP_{t-1} + \sum_{j=0}^{q_1} \delta_i \Delta \ln TO_{t-j} + \sum_{l=0}^{q_2} \varphi_l \Delta \ln CLP_{t-l} + \varepsilon_{it} \end{split}$$



Model 5.

$$\Delta \ln GDP_{it} = \beta_0 + \beta_1 \ln GDP_{t-1} + \beta_2 \ln TO_{t-1} + \beta_3 \ln BSI_{t-1}$$
$$+ \sum_{i=1}^{p} \lambda_i \Delta \ln GDP_{t-1} + \sum_{j=0}^{q_1} \delta_i \Delta \ln TO_{t-j} + \sum_{l=0}^{q_2} \varphi_l \Delta \ln BSI_{t-l} + \varepsilon_{it}$$

Here,  $\Delta$  is the first difference operator; GDP is the logarithm of GDP growth rate per capita; InTO is the logarithm of trade openness; DCB, DCP, BRM, CLP, BSI refers to the logarithm of variables of banking sector depth. Stata14, Eview10 and Gauss10 package programs are used to analyze the model.

## 5. Empirical Results and Discussion

#### 5.1. Descriptive Statistics and Unit Root Tests

#### 5.1.1. Descriptive Statistics

After creating the banking sector depth index through Principal Components analysis, descriptive statistics (mode, median, deviation, etc.) of economic growth, trade openness and banking sector depth variables are determined and the correlation between them are given in Table 4. When Table 4 is examined, it shows that there is a significant variability between the countries.

GDP	то	DCB	DCP	BRM	CLP	BSI
0.81	4.99	-2.75e-07	-1.20e-07	2.22e-07	7.18-08	2.82e-10
1.42	5.01	0.05	0.06	-0.11	0.20	0.06
5.24	6.09	2.06	2.06	2.65	1.76	2.07
-4.36	3.63	-3.13	-3.06	-2.97	-5.01	-3.14
1.70	0.51	1.00	1.00	1.00	0.99	1.00
-0.68	-0.05	-0.55	-0.53	0.10	-2.10	-0.54
2.57	2.55	3.19	3.10	3.48	9.08	3.18
1	-0.13	-0.02	-0.01	-0.04	0.22	-0.02
	1	0.57	0.56	0.48	-0.06	0.56
		1	0.99	0,74	0.18	0.99
			1	0.73	0.19	0.97
				1	-0.03	0.74
					1	0.18
						1
	0.81 1.42 5.24 -4.36 1.70 -0.68 2.57 1	0.81     4.99       1.42     5.01       5.24     6.09       -4.36     3.63       1.70     0.51       -0.68     -0.05       2.57     2.55       1     -0.13       1     -0.13       1     -0.13       1     -0.13       1     -0.13       1     -0.13       1     -0.13	0.81         4.99         -2.75e-07           1.42         5.01         0.05           5.24         6.09         2.06           -4.36         3.63         -3.13           1.70         0.51         1.00           -0.68         -0.05         -0.55           2.57         2.55         3.19           1         -0.13         -0.02           1         0.57         1           1         0.57         1           1         -0.13         -0.02           1         0.57         1           -0.13         -0.02         1           -0.57         1         -0.57	0.81         4.99         -2.75e-07         -1.20e-07           1.42         5.01         0.05         0.06           5.24         6.09         2.06         2.06           -4.36         3.63         -3.13         -3.06           1.70         0.51         1.00         1.00           -0.68         -0.05         -0.55         -0.53           2.57         2.55         3.19         3.10           1         -0.13         -0.02         -0.01           1         0.57         0.56         -0.59           1         0.57         0.56         -0.01           1         0.57         0.56         -0.01           1         0.57         0.56         -0.01           1         0.57         0.56         -0.01           1         -0.02         -0.01         -0.02           1         0.57         0.56         -0.55           1         0.99         -0.14         -0.99           1         -0.01         -0.01         -0.01           1         -0.01         -0.01         -0.01	0.81         4.99         -2.75e-07         -1.20e-07         2.22e-07           1.42         5.01         0.05         0.06         -0.11           5.24         6.09         2.06         2.06         2.65           -4.36         3.63         -3.13         -3.06         -2.97           1.70         0.51         1.00         1.00         1.00           -0.68         -0.05         -0.55         -0.53         0.10           2.57         2.55         3.19         3.10         3.48           -         -         -         -         -           1         -0.13         -0.02         -0.01         -0.04           1         0.57         0.56         0.48           -         1         0.99         0,74           1         -         1         0.73	0.81         4.99         -2.75e-07         -1.20e-07         2.22e-07         7.18-08           1.42         5.01         0.05         0.06         -0.11         0.20           5.24         6.09         2.06         2.06         2.65         1.76           -4.36         3.63         -3.13         -3.06         -2.97         -5.01           1.70         0.51         1.00         1.00         1.00         0.99           -0.68         -0.05         -0.53         0.10         -2.10           2.57         2.55         3.19         3.10         3.48         9.08           -1         -0.13         -0.02         -0.01         -0.04         0.22           1         -0.13         -0.02         -0.01         -0.04         0.22           1         -0.57         0.56         0.48         -0.06           1         0.57         0.56         0.48         -0.06           1         0.57         0.56         0.48         -0.06           1         0.73         0.19         1         -0.03           1         1         0.73         0.19         1           1         1

**Table 4.** Descriptive Statistics and Correlations of Variables

When the correlation between the variables is examined, it shows that the indicators representing the depth of the banking sector are highly correlated with each other and with the created banking sector depth index. This means that if all the variables of the banking sector, economic growth variable and trade openness variable expressed simultaneously in a regression equation, it may cause multiple linear connection problems. Therefore, in this study, the relationship between economic growth and trade openness is examined by considering the variables of banking sector depth separately.



## 5.1.2. The Findings of Unit Root and Stationary Analysis Test in Panel Data for All Countries Group (ACG)

The first-generation unit root tests applied to the ACG group are LLC, ADF, PP and IPS. For all four tests, the null hypothesis H0 indicates that all seven variables (GDP, TO, DCB, DCP, BRM, CLP, BSI) have a unit root (non-stationary), while the alternative hypothesis states that the variables do not have a unit root (stationary). The results of the analysis are shown in Table 5.

Variable	Level		LLC	ADF	PP	IPS
	Loval	Stat.	-5.76*	10.3*	22.3*	-8.30*
CDD	Level	Prob.	0.00	0.00	0.00	0.00
GDP	First Difference (FD)	Stat.	-13.3	48.5	103.3	-14.27
	First Difference (FD)	Prob.	0.00	0.00	0.000	0.00
	Loval	Stat.	-1.22	3.90*	2.85*	-1.33
то	Level	Prob.	0.11	0.00	0.002	0.09
ТО	First Difference (FD)	Stat.	-8.39*	23.4	46.6	-10.81*
	First Difference (FD)	Prob.	0.00	0.00	0.00	0.00
	1	Stat.	-1.09	0.29	-1.31	2.35
DCB	Level	Prob.	0.14	0.38	0.90	0.99
		Stat.	-10.4*	27.9*	32.6*	-9.34*
	First Difference (FD)	Prob.	0.00	0.00	0.00	0.00
	Level	Stat.	-1.37	0.28	-1.35	1.96
DCD		Prob.	0.09	0.39	0.91	0.97
DCP	First Difference (FD)	Stat.	-10.5*	28.4*	33.0*	-9.44*
		Prob.	0.00	0.00	0.00	0.00
		Stat.	-1.94**	7.82*	1.39	1.90
DDM	Level	Prob.	0.03	0.00	0.08	0.97
BRM		Stat.	-5.99	21.1	51.9 *	-10.76*
	First Difference (FD)	Prob.	0.00	0.00	0.00	0.00
	Loval	Stat.	-5.36*	12.9*	28.4*	-9.023*
CLD	Level	Prob.	0.00	0.00	0.00	0.00
CLP	First Difference (FD)	Stat.	-17.5	61.1	97.2	-13.87
	First Difference (FD)	Prob.	0.00	0.00	0.00	0.00
	Lovel	Stat.	-1.09	0.29	-1.31	2.35
BSI	Level	Prob.	0.14	0.39	0.91	0.99
		Stat.	-10.2*	27.7*	32.6*	-9.34
	First Difference (FD)	Prob.	0.00	0.00	0.00	0.00

Note: \* and \*\* represent statistical significance at 1% and 5%, respectively. **Table 5.** Unit Root Test Results of All Countries Group

When the results obtained in Table 5 are examined, it is seen that some of the variables are stationary at level I (0), and some of them are stationary in the first difference I (1).

#### 5.1.3. The Findings of Cross-Sectional Dependence Test

The findings obtained from the first-generation unit root test show that some of the variables used in the study are stationary at level and some of them are stationary in the first difference I (1). In other words, the fact that the variables become stationary at different levels suggests that the cross-sectional dependence problem may have occurred. Because if there is a cross-sectional dependence between the variables, the unit root can lead to rejection of non-stationary null hypothesis (O'Connell, 1998). In this context, the findings in Table 5 suggest that there may be more rejections than expected. In other words, since the analyzes are performed assuming the lack of inter-unit correlation between the variables in first generation root tests, , it is possible to give biased results.



Therefore, the CDLM1 test of Berusch-Pagan (B-P) (1980), one of the cross-sectional dependence tests which test whether there is correlation between units, is given in Table 6 together with Pesaran (2004) CDLM2 tests. In addition, the findings obtained for each model are given in Table 7.

	All Countries G	All Countries Group								
Variables	Tests	Tests								
	LM1 (Breusch,	Pagan)	LM2 (Pesera	in CD)						
	Stat.	Stat. Prob.		Prob.						
GDP	184.690	0.0034834	2.952	0.0015774						
то	253.010	0.0000000	7.095	0.0000000						
DCB	214.391	0.0000206	4.753	0.0000010						
DCP	264.796	0.0000000	7.809	0.0000000						
BRM	420.630	0.0000000	17.258	0.0000000						
CLP	228.068	0.0000012	5.582	0.0000001						
BSI	298.379	0.0000000	9.846	0.0000000						

Table 6. Cross-Sectional Analysis Results of Variables (All Countries Group)

	All Countries C	All Countries Group							
Models	Tests	Tests							
	LM1 (Breusch,	LM1 (Breusch, Pagan) LM2 (Peseran CD)							
	Stat.	Stat. Prob. Stat.							
Model 1	179.563	0.007	2.641	0.004					
Model 2	182.760	0.005	2.835	0.002					
Model 3	168.815	0.029	1.990	0.023					
Model 4	170.693	0.023	2.104	0.018					
Model 5	5179.563	0.007	2.641	0.004					

 Table 7. Cross-Sectional Analysis Results of Models (All Countries Group)

As a result of the test of the cross-sectional dependence of the variables and models given in Table 6 and Table 7, the null hypothesis (H0), which has no cross-sectional dependence, is rejected. Therefore, since it is concluded that there is a cross-sectional dependence, second generation unit root tests are applied.



#### 5.1.4. Second Generation Unit Root Test Results

In Table 8, the critical values in the I (b) and II (b) tables of Pesaran (2007) and the tstatistics obtained from the second-generation unit root tests CADF and CIPS test are given.

	Fixed I	Nodel						Critical	/alues	
Countries	GDP	то	DCB	DCP	BRM	CLP	BSI	0.01	0.05	0.10
Bahrain	3.93 <sup>b</sup>	-4.12 <sup>b</sup>	3.87 <sup>b</sup>	-2.75	-2.45	-3.32°	-2.98	-4.12	-3.36	-2.98
Kuwait	-2.07	-2.55	3.65 <sup>b</sup>	-3.23 <sup>c</sup>	-1.91	-2.55	-3.61 <sup>b</sup>	-4.12	-3.36	-2.98
S. Arabia	-1.80	-2.68	3.03°	-4.10 <sup>b</sup>	-2.63	-1.41	-3.18 <sup>c</sup>	-4.12	-3.36	-2.98
U. Arab Emirates	4.33ª	-2.88	-1.54	-1.75	-3.25°	-3.18 <sup>c</sup>	-4.06 <sup>b</sup>	-4.12	-3.36	-2.98
Qatar	3.32°	-2.86	-2.00	-3.49 <sup>b</sup>	-4.35ª	-2.77	-2.79	-4.12	-3.36	-2.98
Turkey	-2.89	-3.60 <sup>b</sup>	-2.54	-2.60	-2.93	-2.61	-3.12°	-4.12	-3.36	-2.98
Algeria	-2.54	-3.11 <sup>c</sup>	3.00 <sup>c</sup>	-2.64	-1.25	-2.76	-2.95	-4.12	-3.36	-2.98
Lebanon	-2.48	-3.07 <sup>c</sup>	-2.24	-2.23	-3.71 <sup>b</sup>	-2.78	-2.32	-4.12	-3.36	-2.98
Malaysia	-2.88	-2.01	-2.69	-2.61	-2.59	-2.86	-2.68	-4.12	-3.36	-2.98
Bangladesh	-1.96	-3.15 <sup>b</sup>	-1.73	-2.68	-4.00ª	-2.80	-2.41	-4.12	-3.36	-2.98
Egypt	3.40 <sup>b</sup>	-2.22	-1.46	-1.90	-3.02°	-4.84ª	-2.84	-4.12	-3.36	-2.98
Indonesia	-2.85	-2.22	-3.51	-2.44	-2.82	-3.36 <sup>b</sup>	-2.84	-4.12	-3.36	-2.98
Jordan	-2.48	-2.78	-2.32	-2.38	-2.09	-2.05	-2.37	-4.12	-3.36	-2.98
Tunisia	3.31 <sup>c</sup>	-2.36	-2.21	-1.86	-3.33°	-1.53	-1.30	-4.12	-3.36	-2.98
Pakistan	-2.21	-3.39 <sup>b</sup>	3.53 <sup>b</sup>	-3.44 <sup>b</sup>	-1.82	-4.28ª	-2.65	-4.12	-3.36	-2.98
Senegal	-2.93	-3.56 <sup>b</sup>	-1.95	-3.06°	-2.58	-3.87 <sup>b</sup>	-2.06	-4.12	-3.36	-2.98
Gambia	-2.73	-2.87	4.68ª	-3.50 <sup>b</sup>	-1.98	-2.97	-4.99ª	-4.12	-3.36	-2.98
CIPS	2.83ª	-2.91ª	2.70ª	-2.74ª	-2.75ª	-2.93ª	-2.89ª	-2.45	-2.25	-2.14

Note: In the test statistic results a, b, c show statistical significance at 1%, 5% and 10% levels, respectively. In the table above, the individual critical value of each country is based on the Table I (b) mentioned in Pesaran (2007) 's survey on pages 275-276 and the nearest value for N = 17, t = 27 is obtained by taking the values corresponding to N=15, T = 30. In addition, critical values of the overall panel were obtained by looking at Table II (b) in the same study.

**Table 8.** Fixed Model's CADF Test Results at Level (All Countries Group)

Table 8 shows the second generation CADF test for the fixed model in level. In contrast to the null hypothesis that there is a unit root, we test the case where at least one of the series is stationary in the alternative hypothesis. If the CADFcalculated<CADFcritical, the null hypothesis cannot be rejected, and the series is said to have a unit root. When the Table 8. is examined, it can be said that most of the calculated value of each variable of the country data is generally lower than the critical values corresponding to the critical values of Table I (b) of Pesaran (2007) and given in Table 8 and therefore contains unit root. The other part shows a weak stagnation (expressed c in the 10% significance level and related table). However, for each variable, it is found that all the variables for the overall model expressed by CIPS are stationary in the fixed model at the level. The CADF and CIPS second generation unit root test of the fixed and trendy model was performed for all countries. In the second generation CADF and CIPS test for the fixed and trend model at the level, the null hypothesis is "there is a unit root", while the alternative hypothesis "at least one of the series is stationary" is tested. In this test, if the CADF<sub>calculated</sub><CADF<sub>critical</sub>, the null hypothesis cannot be rejected, and the series is said to have a unit root. When the Table 8 is examined, it is concluded that a very important part of the country data except for very little of the calculated value of each variable is generally lower than the critical values corresponding to the critical values of Table I (c) of Pesaran (2007) and therefore contains unit root. The other part shows a weak stationary (in 10% significance level and expressed as "c" in the Table 8). In addition to this, it is founded that for each of the variables expressed by the CIPS model for the general variables BRM, CLP and BSI variable at level and trend model with 1% and 5% significance levels



are not stationary and shows a weak stationarity at the 10% significance level while the other variables shows stationarity at 5% significance level. When the CADF test findings are evaluated in general for the "fixed model" and "fixed and trend model", it is concluded that a significant part of the country data is not stationary, and a small part is stationary. Moreover, when the CIPS values of each variable are examined for the overall model, it is concluded that the data in the fixed model is stationary at 1%, 5% and 10% significance level, but in the "fixed and trend" model some of the variables show weak stationarity and the remaining part is stationary at 5% significance level. Since both stationary and non-stationary results were obtained in both model of variables (fixed model and fixed and trend model), CADF test was applied in both models by taking the first difference. Thus, it was tested whether the variables for the country series and the model in general turned stationary I (1) in the first difference. The fixed model table with the first difference is given in Appendix 1. When the table of first difference of fixed model in Appendix 1 is examined, it is shown that a very important part of all country series reached stationarity at 1% and 5% significance levels, while a few of them were stationary at a 10% significance level. As a result, it is observed that all country series reached stationarity in I (1). On the other hand, in the CIPS test which gives information about the whole model, it is observed that all variables are stationary at a significance level of 1%, in other words, I (1). In short, when the first- and second-generation unit root analysis results are evaluated together, panel Westerlund (2007) cointegration analysis and panel ARDL cointegration analysis were applied for the following reasons;

- As a result of the first generation unit root tests, some of the variables reached stasis in I (0) and some of them reached I (1). For this reason, panel Westerlund (2007) and panel ARDL cointegration analysis, which are suitable for series that do not reach at the same level of stationarity, were used.
- 2. As a result of the second generation unit root tests, which are stronger than the first generation tests and taking into account the cross-sectional dependence, some of the country series and the variable series for the whole of the panel are stationary at level I (0) and the other part is stationary at level (1).

For this reason, Westerlund (2007) error correction cointegration test and panel ARDL cointegration analysis, which is consistent for series that do not reach at the same level of stationarity, have been used (Pesaran and Smith, 1995; Pesaran, Shin and Smith (1999); Gerni et al., 2013). Also, the fact that the series is I (1) is a prerequisite for the cointegration analysis (Koçbulut and Altıntaş, 2016). For the above reasons, Westerlund (2007) and panel ARDL cointegration analysis will be used instead of conventional panel cointegration tests (Pedroni, 1999; Johnsen, 1988). In addition to this, instead of cointegration tests of Pedroni (1999, 2004) and Panel CUSUM of Westerlund (2005), assuming cross-sectional independence, it is preferred to use the Westerlund (2007) and panel ARDL approach. Among these, firstly, Westerlund (2007) error correction based cointegration test was used to determine whether there is cointegration between variables or not. Then, short-term dynamic impact and long-term relationships will be examined using panel data estimators based on the panel ARDL approach.



#### 5.1.5. The Results of Westerlund (2007) Cointegration Test of ACG

Westerlund (2007) uses four test statistics to test the existence of cointegration. For group statistics (Gt and Ga), the null hypothesis is "there is no cointegration for crosssectional units" and the alternative hypothesis is that "there is no cointegration in some units but there are cointegration in some units". Similarly, the null hypothesis of the Pa and Pt test statistics that indicating information for all panel is "no cointegration for all cross-sectional units" and the alternative hypothesis is "cointegration for all cross-sectional units". The Westerlund (2007) cointegration test using the xtwest command in the Stata Program states that the results of the study with small data sets (such as the study with T = 27) may be sensitive to the selection of parameters such as lag, lead and kernel width Westerlund (2007). Therefore, it is recommended that the number of lags and leads be small and the kernel width shorter to avoid excessive parameterization (Westerlund, 2007; Demetriades and James, 2011). In our study, the number of lags was taken as 1 (lag=1) as stated in the following sections. In addition, the number of leads was determined as 1. The kernel width was determined to be approximately 3 (4. (27/100) 2 / 9≈3) with the formula 4 (T / 100) 2/9 as suggested by Persyn and Westerlund (2008). Critical values required for testing these hypotheses are determined with the help of bootstrap cycle (Westerlund, 2008: 200-203). The presence of cointegration in 5 different models for each group of countries in the study was tested separately and the findings are presented in Table 9.

	Test	Value	Z-value	P-value	Robust P-value
	Gt	-2.803	-3.437	0.000	0.000
Model 1: GDP TO DCB	Ga	-10.529	-0.923	0.178	0.000
	Pt	-11.916	-4.699	0.000	0.000
	Pa	-10.810	-3.646	0.000	0.000
	Gt	-2.870	-3.739	0.000	0.000
Model 2: GDP TO DCP	Ga	-10.815	-1.111	0.133	0.000
MOUEL 2. GDP TO DCP	Pt	-11.782	-4.570	0.000	0.000
	Pa	-10.832	-3.663	0.000	0.000
	Gt	-2.807	-3.456	0.000	0.000
Model 3: GDP TO BRM	Ga	-8.788	0.221	0.588	0.020
	Pt	-10.599	-3.424	0.000	0.000
	Pa	-8.397	-1.867	0.031	0.000
	Gt	-2.876	-3.765	0.000	0.000
Model 4: GDP TO CLP	Ga	-10.317	-0.784	0.217	0.000
	Pt	-12.083	-4.861	0.000	0.000
	Pa	-10.539	-3.447	0.000	0.000
	Gt	-2.803	-3.437	0.000	0.000
Model 5: GDP TO BSI	Ga	-10.529	-0.923	0.178	0.000
	Pt	-11.916	-4.699	0.000	0.000
	Pa	-10.810	-3.646	0.000	0.000

**Table 9.** Westerlund (2007) Cointegration Test (All Countries Group)

When the P-value and Robust P-value values of the test statistics obtained in Table 9 are examined, the null hypothesis which is "there is no cointegration" is rejected in almost all five models according to the P-value values of all statistics. Similarly, according to the value of the Robust P-value, for all statistics "there is no cointegration" hypothesis was rejected. Accordingly, cointegration was obtained in all models for all countries. If a long-term relationship is found between the variables as a result of panel cointegration, long and short term relationships can be estimated by various methods. Fully modified least squares (FMOLS) and panel model with



ordinary least squares (PDOLS) can only estimate long-term parameters (Yerdelen, 2013a). PMG, MG and DFE estimators can predict both long and short term parameters. In this context, PMG, MG and DFE estimators are used to obtain long and short term relationships and parameters between the variables in the panel ARDL cointegration model.

#### 5.1.5.1. Analysis of All Countries Group (ACG) by PMG, MG and DFE Estimator

#### Model 1. GDP TO DCB

First, the long and short term relationships of the trade openness (TO) and DCB, one of the banking sector depth measure and GDP variable will be determined. For this purpose, autoregressive distributed lag model (ARDL) was applied by using pooled mean group (PMG), mean group (MG) and dynamic fixed effects (DFE) estimators. For simplicity, the pooled average group, average group, and dynamic constant effects estimators will be expressed as PMG, MG and DFE estimators, respectively. Before applying these three estimator approaches, it is necessary to determine the number of lags of the model. Choosing too much lag may increase the error in estimates, while too little lag may result in lack of information. There are many information criteria procedures to select the appropriate lags. Three commonly used criteria are Schewartz's Bayesion Information Criteria (SBC), Akaike's Information Criteria (AIC), and Hannen and Quinn Information Criteria (HQ). The lag selection in the ARDL model can be performed using a single equation estimate for each panel unit. Thus, by eliminating serial correlation, selecting an appropriate lag sequence also eliminates problems arising from potential endogenity (Lanzafame, 2013). However, especially in the analysis of short-term parameters, it is recommended to assign the same lag sequence to the variable selected and model (Loayza and Ranciere, 2004). Therefore, it is preferred that the variables in the model have the same number of lags. The Schwarz Bayes Criterion (Schwartz, 1978) is used because it performs better than most other alternatives (Maddala, 1992; Mills and Prasad, 1992). Based on this, the optimal number of lags in all models by reference to the SBC information criterion is obtained 1 when taking the maximum number of lags as 3.

Table 10 shows Model 1 results using PMG, MG and DFE estimators of all countries. In addition, by using Hausman test, PMG and MG estimators and PMG and DFE estimators were compared to determine which estimator was more effective than the other two estimators and the results were reported in the same table. If the probability value of Hausman test is less than 0.05, zero hypothesis is rejected, and alternative hypothesis is accepted. PMG estimator maintains long-term coefficients constant, but allows all slope coefficients, error variations and error correction coefficients (adjustment rate) to vary between countries. On the other hand, the MG estimator allows all slope coefficients and error variations to vary between countries in the short and long run. In addition, the DFE estimator uses the assumption of fixed effects when estimating the error correction model and does not calculate by units since it keeps all parameters constant (Yerdelen, 2013b: 244).

In this context, when the results obtained in Table 10 are examined, the statistically significant error correction coefficient (Ø) with negative signs obtained from all three estimators indicates that there is a long-term relationship between GDP variable and TO and DCB variables. This means that even in case of deviations from the equilibrium in case of crisis, it will converge to the equilibrium again. In other words, the error



correction coefficient indicates the speed at which short-term deviations due to nonstationary series reach equilibrium in the next period (Yerdelen, 2013b: 245). For example, according to the PMG estimator, approximately 82% of imbalances in a period will improve in the next period and approach the long-term balance. Furthermore, when the results Table 10 are considered, according to the PMG estimator, it is observed that the TO variable has a positive and statistically significant effect on GDP in the long term and a positive but insignificant effect in the short term. Accordingly, a 1% increase in the TO variable will increase GDP by about 0.84% in the long run. When the effect of DCB variable on GDP was examined, it was found that DCB variable had a positive and significant effect on GDP in the long term and a positive but insignificant effect in the short term. On the other hand, according to the MG estimator, the TO variable is positive on GDP in the long run and negative in the short run, but these effects are insignificant in both periods. However, DCB variable has a significant negative effect a on GDP in the long run but a negative but insignificant effect on GDP in the short term. Moreover, according to the DFE estimator, it was found that the TO variable had a positive and significant effect on the significance level of 1% in the long run, whereas it had a negative and insignificant effect on GDP in the short run. DCB was found to have a negative and significant effect in the long term and a negative and insignificant effect in the short term. In the Hausman test (Yerdelen, 2013b: 255), which tests the homogeneity of long-term coefficients and can be used to choose among estimators, the hypothesis that PMG estimator which is zero hypothesis is more effective than MG estimator was first tested. In the Hausman test, as expected, the PMG estimator with the null hypothesis was more effective than the MG estimator, and the hypothesis could not be rejected, suggesting that PMG is a more effective predictor than MG. This result, as Hsiao et al. (1999) stated, the MG approach is ineffective for small sample and short time series. In addition, the Hausman Test between DFE and PMG estimators rejected the hypothesis that the PMG Estimator with zero hypothesis is more effective than the DFE estimator, which clearly supports the hypothesis that the DFE estimator is more effective than the PMG and MG. In summary, as the DFE estimator is a more effective estimator than the MG and PMG estimators, when we look at the results of the DFE estimator again, 75% of the imbalances in one period will improve in the next period and approach the long-term equilibrium. In addition, it is found that TO has a positive and significant effect on GDP in the long run with 1% significance level and it had a negative and insignificant effect on GDP in the short run. Accordingly, a 1% increase in the TO variable will increase GDP by 1.81% in the long run. The DCB variable has a negative and significant effect on GDP in the long run and a negative and insignificant effect on the short run. Thus, a 1% increase in the DCB variable will reduce GDP by 0.53% in the long run. In this model, the most effective estimator DFE is the result of the Hausman test and in the DFE estimator, since the error correction model is estimated with the assumption of constant effects and all parameters are kept constant, it is not possible to calculate coefficients according to the related model individually (Yerdelen, 2013b: 244).



Dependent unrighter CDD	PMG		MG		DFE	
Dependent variable: GDP	Coefficients	Std. Error	Coefficients	Std. Error	Coefficients	Std. Error
Long Term Coefficients						
ТО	0.842***	0.215	1.371	1.164	1.807***	0.619
DCB	0.255**	0.112	-1.041**	0.487	-0.531**	0.223
Error Correction Coefficient (Ø)	-0.821***	0.069	-0.989***	0.064	-0.747***	0.046
Short Term Coefficients						
ΔΤΟ (-1)	0.148	1.019	-0.261	1.282	-1.036	0.721
ΔDCB (-1)	-0.956	0.857	-0.242	0.618	-0.064	0.341
C (Constant)	-2.577***	0.224	-4.619	6.488	-6.131***	2.330
Number of observations	459		459		459	
Number of countries	17		17		17	
Hausman Test			5.49ª		20.83 <sup>b</sup>	
p-value			0.0643		0.000	

Note: \*\*\*, \*\*, \* indicate the significance level of 1%, 5% and 10%, respectively. The table reports results for pooled mean group (PMG), mean group (MG) and dynamic constant effects (DFE) estimators. The dependent variable is GDP. The first panel (LR) shows long-term effects, while the second panel shows both equilibrium velocity (Ø) and short-term effects (SR). Hausman:

<sup>b</sup> = PMG is a more efficient estimator than DFE under the null hypothesis.

 Table 10. PMG, MG, DFE Test Results for Model 1 (GDP TO DCB) (All Countries Group)

#### Model 2. GDP TO DCP

Secondly, PMG, MG and DFE estimator results were found for Model 2 of all countries group. More specifically, the PMG, MG and DFE estimator results are given in Table 16 to determine the long- and short-term relationships of the TO and DCP variables with the GDP variable. In this context, when results obtained in Table 11 are examined, the error correction coefficient (0) is statistically significant and has a negative sign in all three estimators, indicating that there is a long-term relationship between the GDP, TO and DCP variables, and that shows short-term deviations will converge to the longterm equilibrium. Also, Hausman test gives that the DFE estimator is a more effective estimator than the MG and PMG estimators in the same Table. DFE estimator coefficients in Table 11 are representing that the speed of imbalances in the short term approaching to equilibrium in the following period is high (75%). According to the results obtained in this model, in short, the DFE estimator is a more effective estimator than the MG and PMG estimators, and according to the results of this estimator, TO and DCP variables have positive and significant negative effects on GDP in the long run and significant but insignificant in the short run. More specifically, a 1% increase in the TO variable will increase GDP by 1.79% in the long run. The 1% increase in the DCP variable will reduce GDP by 53% in the long run. In this model, as in Model 1, the most effective estimator DFE is the result of the Hausman test and in the DFE estimator, the coefficient calculations of the relevant model cannot be made according to units (countries) since the error correction model is estimated with the assumption of constant effects and all parameters are kept constant.



<sup>&</sup>lt;sup>a</sup> = PMG is a more effective estimator than MG under the null hypothesis.

Dependent variable: GDP	PMG		MG		DFE		
	Coefficients	Std. Error	Coefficients	Std. Error	Coefficients	Std. Error	
Long Term Coefficients							
ТО	0.918***	0.231	1.292	1.174	1.789***	0.617	
DCP	0.212*	0.122	-1.111**	0.498	-0.526**	0.223	
Error Correction Coefficient (Ø)	-0.820***	0.066	-0.996***	0.065	-0.746***	0.046	
Short Term Coefficients							
ΔΤΟ(-1)	0.045	0.973	-0.251	1.254	-1.038	0.721	
ΔDCP(-1)	-0.749	0.737	-0.129	0.557	-0.081	0.344	
C(Constant)	-2.898***	0.242	-4.062	6.553	-6.060***	2.324	
Number of observations	459		459		459		
Number of countries	17		17		17		
Hausman Test			5.49ª		19.28 <sup>b</sup>		
p- Value			0.0642		0.0001		

Note: \*\*\*, \*\*, \* indicate the significance level of 1%, 5% and 10%, respectively. The table reports results for pooled mean group (PMG), mean group (MG) and dynamic constant effects (DFE) estimators. The dependent variable is GDP. The first panel (LR) shows long-term effects, while the second panel shows both equilibrium speed (Ø) and short-term effects (SR). Hausman: a = PMG is a more effective estimator than MG under the null hypothesis.

b = PMG is a more efficient estimator than DFE under the null hypothesis.

 Table 11. PMG, MG, DFE Test Results for Model 2 (GDP TO DCP) (All Countries Group)

#### Model 3. GDP TO BRM

Table 12 shows the PMG, MG and DFE estimator results for the Model 3 of the all countries groups. In this model, PMG, MG and DFE estimator results were found in order to determine the long and short term relationships of the TO and BRM variables with the GDP variable. In the results in Table 12, since the error correction coefficient (0) has negative signs in all three estimators and it is statistically significant, there is a long-term relationship between the GDP variable and the TO and BRM variables, so that short-term deviations will converge to long-term equilibrium. According to Hausman tests, PMG estimator is more effective estimator than MG and DFE estimators. Also, in Table 12, PMG estimator coefficients indicated that the convergence rate of the imbalances that occur in the short term is high in the following periods (76%). In short, according to the results in this model, PMG estimator is a more effective estimator than MG and DFE estimators, and according to these estimator results, the TO variable increase in TO will increase GDP by 0.74%. When the effect of BRM variable on GDP is examined, it is seen that BRM variable has a positive and significant effect on GDP in the long term and a negative and significant effect in the short term. Accordingly, the 1% increase in the BRM variable will increase GDP by 0.24% in the long term and decrease by 1.92% in the short term. The results of the PMG estimator for the country (unit) for Model 3 are given in Appendix 2. Thus, the error correction parameter is significant and negative in all countries, so it is found that there is a long-term relationship between GDP, TO and BRM variable in all countries. When short-term coefficients are analyzed in the same table, it is seen that only in Bangladesh and Indonesia countries TO variable has a short-term and significant relationship with GDP variable. The fact that both countries have more populations than other countries may be a reason to explain the relationship between GDP and TO as well as the short-term relationship. In addition, the countries where the BRM variable has a short-term and significant relationship with the GDP variable are Kuwait, Qatar, Saudi Arabia, Turkey and the United Arab Emirates. The fact that the BRM (broadly defined ratio of money supply to GDP), one of the banking sector depth variables, is associated with GDP in both the long and short term, may be a



reason to explain this relationship, especially in those countries where participation banking is more advanced than in other countries.

Dependent variable: GDP	PMG		MG		DFE	
Dependent variable: GDP	Coefficients	Std. Error	Coefficients	Std. Error	Coefficients	Std. Error
Long Term Coefficients						
ТО	0.744***	0.238	0.868	1.240	1.008	0.680
BRM	0.244**	0.102	0.0742	0.629	0.030	0.262
Error Correction Coefficient (Ø)	-0.759***	0.063	-0.869***	0.062	-0.727***	0.044
Short Term Coefficients						
ΔΤΟ (-1)	0.490	0.657	0.609	0.906	-0.103	0.701
ΔBRM (-1)	-1.929***	0.542	-2.291***	0.615	-2.486***	0.375
C(Constant)	-2.079***	0.198	-1.808	5.804	-2.991	2.494
Number of observations	459		459		459	
Number of countries	17		17		17	
Hausman Test			0.05ª		0.80 <sup>b</sup>	
p- Value			0.9732		0.6692	

Note: \*\*\*, \*\*, \* indicate the significance level of 1%, 5% and 10%, respectively. The table reports results for pooled mean group (PMG), mean group (MG) and dynamic constant effects (DFE) estimators. The dependent variable is GDP. The first panel (LR) shows long-term effects, while the second panel shows both equilibrium velocity (Ø) and short-term effects (SR). Hausman: a= PMG is a more effective estimator than MG under the null hypothesis.

b= PMG is a more effective estimator than DFE under the null hypothesis.

Table 12. PMG, MG, DFE Test Results for Model 3 (GDP TO BRM) (All Countries Group)

#### Model 4. GDP TO CLP

For Model 4, PMG, MG and DFE estimator results of all group of countries with dual banking are given in Table 13. In this model, PMG, MG and DFE are used to analyze the long and short term relationships between TO, CLP and GDP variables. When we look at the results in Table 13, we find that there is a long-term relationship between the GDP, TO and CLP variables since the error correction coefficient (Ø) has a negative sign and is statistically significant in all three estimators as in previous models. Therefore, this model shows that the deviations that may occur in the short term will converge to the equilibrium in the long term. According to the results of Hausman tests, DFE estimator results are more effective than PMG and MG estimators. When the DFE estimator coefficients in Table 13 are examined again, the rate at which short-term imbalances converge to equilibrium in subsequent periods is quite high (about 77%). The results of this model show that the DfE estimator is a more effective estimator than the PMG and MG estimators, and that the TO variable has a positive effect on GDP in the long term and negative in the short term but these effects are significant in both periods. Accordingly, the 1% increase in the TO in the long term will increase GDP by 1.31% and reduce it by 1.26% in the short term. The CLP variable has a positive and significant effect on GDP in the long run. Thus, a 1% increase in the CLP variable in the long run will increase GDP by 0.24%. In the short term, CLP variable has a positive but insignificant effect on GDP. In this model, as in Model 1 and 2, the most effective estimator DFE is the result of the Hausman test, and in the DFE estimator, since the error correction model is estimated with the constant effects and all parameters are kept constant, it is not possible to calculate coefficients separately according to the units (countries).



	PMG		MG		DFE	
Dependent variable: GDP	Coefficients	Std. Error	Coefficients	Std. Error	Coefficients	Std. Error
Long Term Coefficients				·	·	
ТО	1.336***	0.133	0.184	1.0834	1.314**	0.558
CLP	-0.060	0.0711	0.178	0.265	0.244*	0.137
Error Correction Coefficient (Ø)	-0.757***	0.062	-0.891***	0.063	-0.765***	0.047
Short Term Coefficients						
ΔΤΟ (-1)	-0.722	1.075	-0.118	1.431	-1.256*	0.710
ΔCLP (-1)	0.559**	0.242	0.422*	0.245	0.103	0.077
C(Constant)	-4.422***	0.400	0.803	6.601	-4.408**	2.137
Number of observations	459		459		459	
Number of countries	17		17		17	
Hausman Test			1.77a		7.67b	
p- Value			0.4126		0.0216	

Note: \*\*\*, \*\*, \* indicate the significance level of 1%, 5% and 10%, respectively. The table reports results for pooled mean group (PMG), mean group (MG) and dynamic constant effects (DFE) estimators. The dependent variable is GDP. The first panel (LR) shows long-term effects, while the second panel shows both equilibrium velocity (0) and short-term effects (SR). Hausman:

a= PMG is a more effective estimator than MG under the null hypothesis.

b=PMG is a more effective estimator than DFE under the null hypothesis.

Table 13. PMG, MG, DFE Test Results for Model 4 (GDP TO CLP) (All Countries Group)

#### Model 5. GDP TO BSI

Finally, PMG, MG and DFE estimator results for Model 5 of all group of countries with dual banking have been found. In this model, PMG, MG and DFE estimator results are used to determine long-term and short-term relationships between the trade openness (TO), composite variable of the banking sector depth (BSI) and gross domestic product growth rate (GDP). The results are given in Table 14. Accordingly, when the results obtained in Table 14 are examined, since the error correction coefficient (Ø) is negative and statistically significant for all three estimators, there is a long-term relationship between GDP variable and TO and BSI variables, which may occur in the short term. It shows that deviations will converge to equilibrium in the long term. According to the results of Hausman tests, the DFE estimator is a more effective estimator than the MG and PMG estimators. DFE estimator coefficients in Table 14 also shows that the speed of short-term imbalances approaching equilibrium in the next period is quite high (approximately 75%).) According to the results of this model, in summary, the DFE estimator is a more effective estimator than the MG and PMG estimators, and according to this estimator results, the TO variable has a positive and significant effect on GDP in the long run and a negative and insignificant effect in the short run. Accordingly, a 1% increase in the TO variable will increase GDP by 1.81% in the long run. It is concluded that BSI variable has a negative and significant effect on GDP in the long run and a negative and insignificant effect on the short run. Accordingly, a 1% increase in the BSI variable over the long term will reduce GDP by about 0.53%. This shows how closely the composite index derived from the other four variables is related to GDP. In this model, as in Models 1 and 2 and 4, the most effective estimator DFE is the result of the Hausman test, and in the DFE estimator, since the error correction model is estimated with the constant effects and all parameters are kept constant, it is not possible to calculate coefficients individually according to the units (countries).



	DIAG				DEE	
Dependent variable: GDP	PMG		MG		DFE	
	Coefficients	Std. Error	Coefficients	Std. Error	Coefficients	Std. Error
Long Term Coefficients						
ТО	0.843***	0.215	1.371	1.164	1.807***	0.619
BSI	0.255**	0.113	-1.043**	0.487	-0.533**	0.223
Error Correction Coefficient (Ø)	-0.821***	0.069	-0.989***	0.064	-0.747***	0.046
Short Term Coefficients						
ΔΤΟ(-1)	0.148	1.019	-0.261	1.282	-1.036	0.721
ΔBSI(-1)	-0.957	0.858	-0.242	0.619	-0.065	0.342
C(Constant)	-2.577***	0.224	-4.619	6.488	-6.131***	2.330
Number of observations	459		459		459	
Number of countries	17		17		17	
Hausman Test			5.49a		20.83b	
p- Value			0.0643		0.000	

Note: \*\*\*, \*\*, \* indicate the significance level of 1%, 5% and 10%, respectively. The table reports results for pooled mean group (PMG), mean group (MG) and dynamic constant effects (DFE) estimators. The dependent variable is GDP. The first panel (LR) shows long-term effects, while the second panel shows both equilibrium velocity (Ø) and short-term effects (SR). Hausman: a= PMG is a more effective estimator than MG under the null hypothesis.

b= DFE is a more effective estimator than PMG under the null hypothesis.

 Table 14. PMG, MG, DFE Test Results for Model 5 (GDP TO BSI) (All Countries Group)

When the results obtained within the framework of the relevant models for the TUG countries group are evaluated in general, it is concluded that all the BSD variables have a significant long-term relationship with GDP. This clearly shows that in the sample of all countries with dual banking, the variables of the banking sector depth have a long-term relationship with economic growth. Especially in the third model where the BRM variable is one of the BSD variables, the fact that there is a significant relationship between GDP and BRM variables in the short term shows the importance of this variable in terms of its relationship to economic growth. In addition, as a result of the Hausman test, it was observed that the DFE estimator was the most effective model and, in these models, there is also a long-term relationship between the TO variable and GDP and no short-term relationship. In the third model, where the PMG estimator is the effective model, there is a long-term and significant relationship between TO variable and GDP, but no short-term relationship has been found. In this model, the BRM variable has a positive and significant relationship with GDP in the long term, but a negative relationship in the short term. However, the deviations that may occur in the short term show that the model will converge to equilibrium in the long term due to the negative and significant output of the model. In the composite index model, there is no significant difference, but it is observed that it gives similar results with the banking sector depth variables in other models.

## 6. Empirical Results and Discussion

The relationship between economic growth, trade openness and financial development has been and remains the focus of interest for many academic studies. In this study, firstly, it is aimed to define the concepts of banking sector depth, economic growth and trade openness, to examine them in a theoretical framework and to investigate the relationship between them. Then, as the main purpose of the study, it is aimed to examine the relationship between banking sector depth, economic growth and trade openness by panel data analysis method with using annual data covering 1990-2016 of 17 countries that are members of the Organization of Islamic Cooperation of which has dual banking (conventional and participation banking) system. In this context, selected variables that can represent



economic growth, trade openness and banking sector depth have been tested within the framework of determined models. However, it is aimed to carry out the research in a much wider scope by considering the two dimensions of financial depth, banking sector depth and stock sector depth. However, due to the fact that the stock markets of some countries have been established very recently (for example the United Arab Emirates) and some countries do not have a sufficiently long time series (Naceur et al., 2014), the study was carried out only by taking into account the variables of banking sector depth. However, in the relevant sample, a significant part of the financial depth or development and the basis of the financial market is based on the banking sector compared to other countries (Aliyu et al., 2017; Hussain et al., 2015; Kammer et al., 2015). It is of great importance in terms of reflecting the overall financial system in the sample. As the methodology followed in the study, a variable called the banking sector depth composite index (BSI) was derived through the analysis of principal components by using four banking sector depth variables. Then, the first generation unit root tests Levin-Lin-Chu (LLC), Im-Pesaran-Shin (IPS), ADF – Fisher chi-square and PP – Fisher Chi-square tests were applied to determine whether they were stationary or not. Then, the CDLM1 test of Berusch-Paqan (B-P) (1980) and Pesaran (2004) CDLM2 tests are applied to determine whether there is a crosssectional dependence among the data. Since the correlation between the units was determined as a result of the cross-sectional dependence test (there was a crosssectional dependence), CADF and CIPS tests which are the second generation unit root tests considering the cross-sectional dependence is applied the data and models. As a result of both first generation unit root tests and second generation unit root tests, which are stronger than first generation tests and take into account crosssectional dependence, some of the data were observed to be I(0) and some to I(1). Therefore, it is found appropriate to apply Westerlund (2007) error correction-based cointegration test and panel ARDL cointegration analysis to the models. After the determination of cointegration between the models established by Westerlund (2007) cointegration test, three estimators based on the panel ARDL model are used to calculate long-term coefficients and determine the long-and short-term cointegration relationship between variables. Thus, using a panel ARDL model based on the use of Pesaran and Smith's mean group estimator (MG), pooled mean group estimator (PMG) and dynamic fixed effects (DFE) estimator of the three alternative estimators, the long and short term relationships between the banking sector depth (BSD), economic growth and trade openness (TO) is examined. When the results are evaluated in general, it is concluded that all banking sector depth (BSD) variables have a significant long-term relationship with GDP. This clearly shows that the depth of the banking sector in the group of countries with dual banking has a long-term relationship with economic growth. particularly, in the third model, where there is a broad money supply (BRM) variable from BSD, the fact that there is a significant relationship between GDP and BRM variables in the short term shows the importance of this variable in terms of its relationship to economic growth. Moreover, as a result of the Hausman test, it was observed that the DFE estimator is the most effective model and, in these models, there is also a long-term relationship between the TO variable and GDP and no short-term relationship. In the third model, where the PMG estimator is the effective model, although there is a long-term and significant relationship between the TO variable and GDP, no significant short-term relationship has been found. In this model, it is concluded that BRM has a positive and significant



relationship with GDP in the long term, but a negative relationship in the short term. However, it shows that the short-term deviations will converge to equilibrium in the long-term due to the negative and significant output of the model. In the composite index model, it is observed that there is no significant difference but that it gives similar results with the other banking sector depth variables in models. The positive correlation between DCP (Domestic credit to the private sector) and long-term economic growth is like the study result of Al-Moulani (2016). However, Barajas et al. (2013b) shows that there is a negative relationship between DCP and GDP. This may be due to the sample and other control variables discussed in our study. To summarize, the findings of the link between the depth of the banking sector variables and economic growth suggest that the banking sector generally encourages longterm economic growth in this group of countries. For this reason, it can be stated that a more competitive, efficient and stable banking sector in these countries may increase the economic growth by encouraging the deepening of the banking In future studies, especially when there is enough and healthy data range in participation banking, the relationship between participation and conventional banking with economic growth can be examined comparatively, which of these types of banking is superior to the other and whether this advantage differs from country to country. In this study, the lack of access to the data of twenty-two countries with all dual banking is one of the constraints that prevent the research from being more inclusive and more reflecting the finance-growth relationship.

## References

- Abu-Bader, S., Abu-Qarn, A.S. Financial development and economic growth: empirical evidence from six MENA countries. Rev. Dev. Econ., 2008, 12 (4), 803–817.
- Ahmed, S.M., Ansari, M.I., 1998. Financial sector development and economic growth: the South-Asian experience. J. Asian Econ. 9 (3), 503–517.
- Al Mamun, M., Sohog, K., & Akhter, A.: A dynamic panel analysis of the financial determinants of CSR in bangladeshi banking industry. Asian Economic and Financial Review, 2013, 3(5), 560-578.
- Aliu, A., & Abazi, S.:Financial deepening and economic growth in the Western Balkans, 2015.
- Aliyu, S., Hassan, M. K., Mohd Yusof, R., & Naiimi, N.: Islamic banking sustainability: a review of literature and directions for future research. Emerging Markets Finance and Trade, 2017, 53(2), 440-470.
- Al-Moulani, A. J.:Banking sector depth & long-term economic growth in the GCC States: relationship nature, sector development status & policy implications, (Doctoral dissertation, Department of Economics), 2016.
- Ang, J. B. (2010). Determinants of private investment in Malaysia: what causes the postcrisis slumps?. Contemporary Economic Policy, 28(3), 378-391.
- Ang, J.B., McKibbin, W.J., 2007. Financial liberalization, financial sector development and growth: evidence from Malaysia? J. Dev. Econ. 84 (1), 215–233.
- Arcand, J. L., Berkes, E., & Panizza, U.: Too much finance? IMF Working Paper 12/161, 2012.
- Arestis, P., Demetriades, P. O., & Luintel, K. B.:Financial development and economic growth: the role of stock markets. Journal of money, credit and banking, 2001, 16-41.
- Arteta, C, Eichengreen, B. & Wyplosz, C.:When Does Capital Account Liberalization Help More Than It Hurts?. Center for Economic Policy Research Discussion Paper, No. 2910, 2001.
- Atje, R., & Jovanovic, B. Stock markets and development. European Economic Review, 1993, 37(2-3), 632-640.
- Bagehot, W.: Lombard Street: A description of the money market. Scribner, Armstrong & Company, 1873.
- Bahmani-Oskooee, M., & Niroomand, F.:Openness and economic growth: an empirical investigation. Applied Economics Letters, 1999, 6(9), 557-561.



- Barajas, M. A., Beck, T., Dabla-Norris, M. E., & Yousefi, M. R.: Too cold, too hot, or just right? Assessing financial sector development across the globe (No. 13-81). International Monetary Fund, 2013a.
- Barajas, M. A., Chami, M. R., & Yousefi, M. R.:The finance and growth nexus re-examined: Do all countries benefit equally? (No. 13-130). International Monetary Fund, 2013b.
- Barro, Robert J.:Economic Growth in a Cross Section of Countries. Quarterly Journal of Economics, 1991, 106, 407–41.
- Beck, T., Levine, R. & Loayza, N.:Finance and the Sources of Growth. Journal of Financial Economics. 2000, 58, 261–300.
- Beck, T., Levine, R.:Stock markets, banks and growth: panel evidence. J. Bank. Finance 2004, 28(3), 423–442.
- Beck, T. Finance and Oil: Is there a resource curse in financial development? European Banking Center Discussion Paper No. 2011-004, European Banking Center, Tilburg University, Tilburg, Netherlands, 2011.
- Bencivenga, V. R., Smith, B. D., & Starr, R. M.:Transactions costs, technological choice, and endogenous growth. Journal of economic theory, 1995, 67(1), 153-177.
- Benhabib, J., & Spiegel, M. M.:The role of financial development in growth and investment. Journal of economic growth, 2000, 5(4), 341-360.
- Bhattacharyya, S., & Hodler, R. Do natural resource revenues hinder financial development? The role of political institutions. World Development, 2014, 57, 101-113.
- Blackburn, K., & Hung, V. T.: A theory of growth, financial development and trade. Economica, 1998, 65(257), 107-124.
- Boulila, G., & Trabelsi, M.:The causality issue in the finance and growth nexus: empirical evidence from Middle East and North African countries. Review of Middle East Economics and Finance, 2004, 2(2), 123-138.
- Boutros-Ghali, B. (1999). Unvanquished, a United Nations-United States Saga. IB Tauris.
- Breusch, T.S. ve Pagan, A.R.:The Lagrange Multiplier Test and Its Applications to Modelspecification Tests in Econometrics. Review of Economic Studies, 1980, 47(1), 239-53.
- Chandavarkar, A.:Of finance and development: neglected and unsettled questions. World development, 1992, 20(1), 133-142.
- Chang, T., Simo-Kengne, B. D., & Gupta, R.:The causal relationship between imports and economic growth in the nine provinces of South Africa: evidence from panel Granger causality tests. Journal of Economic Cooperation and Development, 2013, 35(2), 71-90.
- Christopoulos, D. K., & Tsionas, E. G. Financial development and economic growth: evidence from panel unit root and cointegration tests. Journal of development Economics, 2004, 73(1), 55-74.
- Craigwell, R., Downes, D., Howard, M., 2001. The finance-growth nexus: a multivariate VAR analysis of a small open economy. Savings Dev. 25 (2), 209–223
- Demetriades, P. O., & James, G. A. Finance and growth in Africa: The broken link. Economics Letters, 2011, 113(3), 263-265.
- Dollar, D. Outward-oriented developing economies really do grow more rapidly: evidence from 95 LDCs, 1976-1985. Economic development and cultural change, 40(3), 523-544, 1992.
- Dornbusch, R.:Exchange rates and fiscal policy in a popular model of international trade. The American Economic Review, 1975, 65(5), 859-871.
- Dritsakis, N., Adamopoulos, A., 2004. Financial development and economic growth in Greece: an empirical investigation with granger causality analysis. Int. Econ. J. 18 (4), 547–559.
- Edwards, S.:Financial openness, sudden stops, and current-account reversals. American Economic Review, 2004, 94(2), 59-64.
- Edwards, S.:Openness, productivity and growth: what do we really know?. The economic journal, 1998, 108(447), 383-398.
- Gautam, T. K., & Paudel, K. P.: The demand for natural gas in The Northeastern United States. Energy, 2018, 158, 890-898.
- Gerni, C., Kabadayi, B., Yurttancikmaz, Z. C., & Emsen, O. S. The analysis of competitiveness and economic growth: A case study of transition countries. International Business Research, 2013, 6(5), 117.



Ghatak, S., Milner, C., & Utkulu, U.:Trade liberalisation and endogenous growth: Some evidence for Turkey. Economics of Planning, 1995, 28(2-3), 147-167.

Goldsmith, R.: Financial Structure and Development. New Haven, CT: Yale University Press, 1969.

- Greenaway, D. and D. Sapsford (1994). What does liberalisation do for exports and growth? Weltwirtschaftliches Archiv, 130, 152–74.
- Greenwood, J., & Jovanovic, B. Financial development, growth, and the distribution of income. Journal of political Economy, 1990, 98(5, Part 1), 1076-1107.
- Greenwood, J., Smith, B., 1997. Financial markets in development, and the development of financial markets. J. Econ. Dyn. Control 21 (1), 145–181.
- Gries, T., Kraft, M., Meierrieks, D., 2009. Linkages between financial deepening, trade openness, and economic development: causality evidence from Sub-Saharan Africa. World Dev. 37 (12), 1849–1860.
- Hansson, P., & Jonung, L.:Finance and economic growth: the case of Sweden 1834–1991. Research in Economics-Ricerche Economiche, 1997, 51(3), 275-301.
- Harrison, A.:Openness and growth:A time-series, cross-country analysis for developing countries. Journal of development Economics, 1996, 48(2), 419-447.
- Hermes, N., & Lensink, R.:Foreign direct investment, financial development and economic growth. The Journal of Development Studies, 2003, 40(1), 142-163.
- https://data.worldbank.org/products/wdi
- Huang, H. C., & Lin, S. C.:Non-linear finance–growth nexus: A threshold with instrumental variable approach 1. Economics of Transition, 2009, 17(3), 439-466.
- Hussain, M., Shahmoradi, A., Turk, R.:An Overview of Islamic Finance (No. WP/15/120). IMF Working Papers, IMF Working Paper Series, Washington, D.C. USA, 2015.
- J. Schumpeter, The Theory of Economic Development Harvard University Press, Cambridge
- Johnson, H. G.:The monetary approach to the balance of payments: A nontechnical guide. Journal of international economics, 1977, 7(3), 251-268.
- Jung, W. S.:Financial development and economic growth: international evidence. Economic Development and cultural change, 34(2), 333-346, 1986.
- Kammer, A., Norat, M., Pi<sup>~</sup> nón, M., Prasad, A., Towe, C.:Islamic Finance: Opportunities, Challenges, and Policy Options (No. SDN/15/05). Islamic finance, IMF Staff Discussion Note, Washington, D.C. USA, 2015.
- Kar, M., Nazlioglu, S., Agir, H., 2011. Financial development and economic growth nexus in the MENA countries: bootstrap panel granger causality analysis. Econ. Model. 28 (1–2), 685–693.
- Karabıyık, C., & Taşkın, F. D.:The relationship between financial development and economic growth: evidence from Brics countries and Turkey. Trakya Üniversitesi İktisadi ve İdari Bilimler Fakültesi E-Dergi, 2016, 5(2),121-138.
- King, R. G., & Levine, R.: Finance, entrepreneurship and growth. Journal of Monetary economics, 1993b, 32(3), 513-542.
- King, R. G., & Levine, R.:Finance and growth: Schumpeter might be right. The quarterly journal of economics, 1993a, 108(3), 717-737.
- Koçbulut, Ö., & Altıntaş, H.İkiz Açıklar ve Feldstein-Horioka Hipotezi: OECD Ülkeleri Üzerine Yatay Kesit Bağımlılığı Altında Yapısal Kırılmalı Panel Eşbütünleşme Analizi. Erciyes Üniversitesi İktisadi ve İdari Bilimler Fakültesi Dergisi, 2016, (48), 145-174.
- Kök, R., İspir, M. S., & Arı, A. A.: Zengin Ülkelerden Azgelişmiş Ülkelere Kaynak Aktarma Mekanizmasının Gerekliliği ve Evrensel Bölüşüm Parametresi Üzerine Bir Deneme. Uluslararası Ekonomi Konferansı, Türkiye Ekonomi Kurumu, Kıbrıs, 2010.
- Kugler, P. (1991), "Growth, exports and cointegration: an empirical investigation", Weltwirtschafftliches Archiv, Vol. 127 No. 1, pp. 73-82
- Lanzafame, M.:The balance of payments-constrained growth rate and the natural rate of growth: new empirical evidence. Cambridge Journal of Economics, 2013, 38(4), 817-838.
- Law, S. H., & Singh, N. Does too much finance harm economic growth? Journal of Banking & Finance, 2014, 41, 36-44.
- Levine, R., & Renelt, D. A sensitivity analysis of cross-country growth regressions. The American economic review, 1992, 82(4), 942-963.



- Levine, R., Loayza, N. & Beck, T.: Financial intermediation and growth: Causality and causes. Journal of Monetary Economics. 2000, 46, 31–77.
- Levine, R., Zervos, S.:Stock markets, banks and economic growth. Am. Econ. Rev. 1998, 88 (3), 537– 558.
- Levine, R.:Stock Markets, Growth, and Tax Policy. Journal of Finance, 1991, 46, 1445-1465.
- Loayza, N., & Ranciere, R.: Financial development, financial fragility, and growth. The World Bank, 2004.
- Maddala, G.S.: Introduction to econometrics Prentice Hall, New Jersey (1992).
- Manteli, A.:Does Trade Openness Cause Growth? An Empirical Investigation. Master Thesis, Department of Economics, Södertörns University, 2015.
- Matsuyama, K.: Agricultural productivity, comparative advantage, and economic growth. Journal of economic theory, 58(2), 317-334, 1992.
- McKinnon, Ronald I.: Money and Capital in Economic Development. Washington, DC: The Brookings Institution, 1973.
- Mills, J. A., & Prasad, K.: A comparison of model selection criteria. Econometric reviews, 1992, 11(2), 201-234.
- Mohamed, S.E.: Finance-Growth Nexus in Sudan: Empirical Assessment Based on an Application of the Autoregressive Distributed Lag (ARDL) Model. In: M. E. Postalcı (ed.). Üçüncü Uluslararası Öğrenci Konferansı "Social Sosyal Bilimlerde Ampirik Modelleme". İzmir Ekonomi Üniversitesi Yayınları, İzmir, Türkiye, s. 47–66, 2008.
- Naceur, S. B., Cherif, M., & Kandil, M.: What drives the development of the MENA financial sector? Borsa Istanbul Review, 2014, 14(4), 212-223.
- Naceur, S.B., Ghazouani, S.:Stock markets, banks, and economic growth: empirical evidence from the MENA region. Res. Int. Business Finance, 2007, 21 (2), 297-315.
- Neusser, K., & Kugler, M.:Manufacturing growth and financial development: evidence from OECD countries. Review of economics and statistics, 1998, 80(4), 638-646.
- Nili, M., & Rastad, M.:Addressing the growth failure of the oil economies: The role of financial development. The Quarterly Review of Economics and Finance, 2007, 46(5), 726-740.
- O'Connell P.: The overvaluation of purchasing power parity. Journal of International Economics, 1998, 44 (1), 1–19.
- Pagano, M.:Financial markets and growth. An overview. European Economic Review, 1993, 37, 613–622.
- Pedroni, P.: Critical values for cointegration tests in heterogeneous panels with multiple regressors. Oxford Bulletin of Economics and statistics, 1999, 61(S1), 653-670.
- Pedroni, P.: Panel cointegration: asymptotic and finite sample properties of pooled time series tests with an application to the PPP hypothesis. Econometric theory, 2004, 20(3), 597-625.
- Persyn, D., & Westerlund, J.: Error-correction-based cointegration tests for panel data. Stata Journal, 2008, 8(2), 232-241.
- Pesaran, H. & Smith, R. Estimating Long-Run Relationships from Dynamic Heterogeneous Panels. Journal of Econometrics, 1995, 68 (1), 79-11.
- Pesaran, M. H., Shin, Y., & Smith, R. P.:Pooled mean group estimation of dynamic heterogeneous panels. Journal of the American Statistical Association, 1999, 94(446), 621-634.
- Pesaran, M. H.: A simple panel unit root test in the presence of cross-section dependence. Journal of applied econometrics, 2007, 22(2), 265-312.
- Pesaran, M. H.:General diagnostic tests for cross section dependence in panels. Cambridge Working Papers in Economics No. 0435, 2004.
- Pesaran, M.H. ve Shin, Y. (1999). An Autoregressive Distributed Lag Modelling Approach to Cointegration Analysis", Faculty of Economics, University of Cambridge, Cambridge Working Papers in Economics, Working Paper No: 9514, ssp. 1-24.
- Pigka-Balanika V.: The impact of trade openness on economic growth Evidence in Developing Countries. Erasmus School of Economics, 1-32, 2013.
- Pradhan, R. P., Arvin, M. B., Hall, J. H., & Norman, N. R.: ASEAN economic growth, trade openness and banking-sector depth: The nexus. Economia, 18(3), 359-379, 2017a.



- Puryan, V.:The Causal Relationship between Economic Growth, Banking Sector Development and Stock Market Development in Selected Middle-East and North African Countries. International Journal of Economics and Financial Issues, 2017, 7(3), 575-580.
- Reppas, P.A. and Christopoulos, D.K. (2005), "The export-output growth nexus: evidence from African and Asian countries", Journal of Policy Modelling, Vol. 27 No. 8, pp. 929-940.
- Ricardo, D. (1951). 1973. The works and correspondence of David Ricardo, 8.
- Rivera-Batiz, L., Romer, P., 1991. Economic integration and endogenous growth. Q. J. Econ. 106, 531-556.
- Romer, C., & Romer, D. (1998). Monetary policy and the well-being of the poor. Income inequality: Issue and Policy Options (pp. 159-201). Kansas, MO: Federal Reserve Bank of Kansas City.
- Romer, D., 1998. A new assessment of openness and inflation: reply. Q. J. Econ. 113 (2), 649–652
- Romer, P. M. (1990). Endogenous technological change. Journal of political Economy, 98(5, Part 2), S71-S102.
- Rousseau, P. L., & Sylla, R.: Financial systems, economic growth, and globalization. In Globalization in historical perspective (pp. 373-416). University of Chicago Press, 2003.
- Rousseau, P. L., & Wachtel, P. Financial intermediation and economic performance: historical evidence from five industrialized countries. J. Money Credit Bank, 1998, 34 (4), 657–678.
- Rousseau, P. L.: Finance, investment, and growth in Meiji-era Japan. Japan and the World Economy, 1999, 11(2), 185-198.
- Sachs, J. D., Warner, A., Åslund, A., & Fischer, S.: Economic reform and the process of global integration. Brookings papers on economic activity, 1995(1), 1-118, 1995.
- Samargandi, N.: Essays on financial development and economic growth (Doctoral dissertation, Brunel University London), 2015.
- Sarkar, P. Trade openness and growth: is there any link? Journal of economic issues, 2008, 42(3), 763-785.

Schwartz, R. Estimating the dimension of a model. Annals of Statistics, 1978, 6, pp. 461-464.

Smith, A. (1937). The wealth of nations [1776].

- Solarin, S. A., & Shahbaz, M.: Natural gas consumption and economic growth: The role of foreign direct investment, capital formation and trade openness in Malaysia. Renewable and Sustainable Energy Reviews, 2015, 42, 835-845.
- Uslu, E.:Reel Kurun Denge Değerinden Sapmasında Balassa-Samuelson Etkisi: Türkiye Örneği, Yayımlanmamış Uzmanlık Yeterlilik Tezi, Türkiye Cumhuriyet Merkez Bankası, 2012.
- Wachtel, P., and P. Rousseau (1995). Financial Intermediation and Economic Growth: A Historical Comparison of the U.S., U.K. and Canada. In M. Bordo and R. Sylla (eds.), Anglo-American Financial Systems: Institutions and Markets in Twentieth Century North America and the United Kingdom. Homewood, Ill.: Business One Irwin.
- Wachtel, P.: Growth and Finance: What do we know and how do we know it? International Finance, 2001, 4(3), 335-362.
- Walter, J., Baek, J., & Koo, W. W.: International trade and macroeconomic dynamics: The case of US bilateral trade with G-7 countries. Research in Economics, 2012, 66(4), 398-405.
- Westerlund, J.: Testing for error correction in panel data. Oxford Bulletin of Economics and statistics, 2007, 69(6), 709-748.
- Yanikkaya, H.:Trade openness and economic growth: a cross-country empirical investigation. Journal of Development economics, 2003, 72(1), 57-89.
- Yapraklı, S.: Ticari ve finansal dışa açıklık ile ekonomik büyüme arasındaki ilişki: Türkiye üzerine bir uygulama. Ekonometri ve İstatistik e-Dergisi, 2007, (5), 67-89.
- Yerdelen Tatoğlu, F.: İleri panel Veri Analizi: Stata Uygulamalı (2. Baskı). İstanbul: Beta Yayınları, 2013b.
- Yerdelen Tatoğlu, F.: Panel Veri Ekonometrisi: Stata Uygulamalı (2. Baskı). İstanbul: Beta yayınları, 2013a.
- Zaher, T. S., & Kabir Hassan, M.: A comparative literature survey of Islamic finance and banking. Financial Markets, Institutions & Instruments, 2001, 10(4), 155-199.



## Appendix

	Fixed Model							Critical Values		
Bahrain	GDP	ТО	DCB	DCP	BRM	CLP	BSI	0.01	0.05	0.10
Kuwait	-4.50a	-3.91b	-3.89b	-4.28a	-5.00a	-3.84b	-4.23a	-4.12	-3.36	-2.98
Saudi Arabia	-3.84b	-3.92b	-5.29a	-5.53a	-4.89a	-4.15a	-5.55a	-4.12	-3.36	-2.98
Arab Emirates.	-4.92a	-4.14a	-3.66b	-3.94b	-4.65a	-4.01b	-4.13a	-4.12	-3.36	-2.98
Qatar	-3.96b	-4.23a	-3.21c	-4.19a	-6.51a	-5.36a	-6.60a	-4.12	-3.36	-2.98
Turkey	-4.07b	-4.19a	-4.57a	-4.07b	-7.19a	-5.23a	-5.28a	-4.12	-3.36	-2.98
Algeria	-4.27a	-3.34c	-3.89b	-3.99b	-5.06a	-3.87b	-4.53a	-4.12	-3.36	-2.98
Lebanon	-4.48a	-3.35c	-3.85b	-3.92b	-3.95b	-4.13a	-3.98b	-4.12	-3.36	-2.98
Malaysia	-4.06b	-4.06b	-4.77a	-4.79a	-4.80a	-4.27a	-4.43a	-4.12	-3.36	-2.98
Bangladesh	-3.79b	-4.33a	-4.70a	-4.43a	-4.20a	-4.26a	-4.75a	-4.12	-3.36	-2.98
Egypt	-3.64b	-4.03b	-4.77a	-4.85a	-5.27a	-4.00b	-4.30a	-4.12	-3.36	-2.98
Indonesia	-4.35a	-5.31a	-3.93b	-3.62b	-3.63b	-5.25a	-4.84a	-4.12	-3.36	-2.98
Jordan	-4.41a	-4.60a	-5.08a	-5.72a	-5.14a	-5.03a	-4.80a	-4.12	-3.36	-2.98
Tunisia	-4.14a	-5.92a	-5.12a	-5.05a	-4.60a	-4.59a	-4.83a	-4.12	-3.36	-2.98
Pakistan	-5.67a	-4.36a	-4.24a	-5.34a	-4.91a	-4.58a	-6.63a	-4.12	-3.36	-2.98
Senegal	-4.59a	-4.29a	-4.14a	-5.21a	-4.59a	-4.57a	-4.54a	-4.12	-3.36	-2.98
Gambia	-4.52a	-4.81a	-4.76a	-4.66a	-4.60a	-3.83b	-5.80a	-4.12	-3.36	-2.98
Bahrain	-3.39b	-3.74b	-3.39b	-3.88b	-6.34a	-3.72b	-5.67a	-4.12	-3.36	-2.98
CIPS	-4.27a	-4.09a	-4.08a	-4.32a	-4.96a	-4.39a	-4.99a	-2.45	-2.25	-2.14

Note: In the test statistic results a, b, c shows statistical significance at 1%, 5% and 10% levels, respectively. In the table above, the individual critical value of each country is based on the Table I (b) mentioned in Pesaran (2007) 's survey on pages 275-276 and the nearest value for N = 17, t = 27 is obtained by taking the values corresponding to N=15, T = 30. In addition, critical values of the overall panel were obtained by looking at Table II (b) in the same study.

Appendix 2. PMG Estimator Results of Each Country for Model 3 (GDP TO BRM)(All Countries Group)

	ТО		BRM		
	Coefficients	Std. Deviation	Coefficients	Std. Deviation	Error Correction Parametre
Algeria	2.908	3.211	0.814	1.176	-0.545 ***
Bahrain	-1.412	2.576	-0.683	1.758	-0.716***
Bangladesh	0.816**	0.392	-0.465	0.321	-1.194***
Egypt	-2.088	1.478	-0.356	1.357	-0.688***
Indonesia	-4.293***	1.083	-2.938	2.190	-0.259*
Jordan	2.185	3.588	-1.753	2.312	-0.477***
Kuwait	-1.927	4.217	-3.213***	1.071	-0.603***
Lebanon	2.460	2.249	-3.329	2.091	-0.494***
Malaysia	1.391	4.942	-0.728	1.765	-0.955***
Pakistan	-2.838	2.680	1.282	1.804	-0.729***
Qatar	5.018	5.392	-3.315***	1.273	-1.161***
Saudi Arabia	3.082	4.145	-4.263**	2.020	-0.845***
Senegal	-1.488	2.546	0.760	1.808	-0.890***
Gambia	-0.428	2.558	0.275	1.644	-1.018***
Tunisia	2.358	2.777	-3.701	3.516	-0.785***
Turkey	4.224	2.840	-5.970***	1.329	-1.000***
U. Arab Emirates	-1.640	5.435	-5.214***	1.844	-0.544***

Note: \*\*\*, \*\* and \* represent significance levels of 1%, 5% and 10%, respectively.

